ADENYL CYCLASE AND ITS RELATIONSHIP TO INSECT DIAPAUSE IN THE EUROPEAN CORN BORER, <u>OSTRINIA NUBILALIS</u> (HÜBNER)

by

Dale Berkman Gelman

Dissertation submitted to the Faculty of the Graduate School of the University of Maryland in partial fulfillment of the requirements for the degree of Doctor of Philosophy 1978

cop 1

APPROVAL SHEET

Title of Dissertation:

Adenyl Cyclase and Its Relationship to Insect Diapause in the European Corn Borer, <u>Ostrinia</u> <u>Nubilalis</u> (Hübner)

Name of Candidate:

Dale Berkman Gelman Doctor of Philosophy, 1978

Dissertation and Abstract Approved:

J. David Beckard

Dr./J. David Lockard Professor Botany and Science Teaching

2. Hayes ora 9

Dr. Dora K. Hayes Laboratory Chief Chemical and Biophysical Control Laboratory AEQI, SEA

Nov, 16, 1978

Date Approved:

ABSTRACT

Title of Dissertation: Adenyl Cyclase and Its Relationship to Insect Diapause in the European Corn Borer, <u>Ostrinia Nubilalis</u> (Hübner)

Dale B. Gelman, Doctor of Philosophy, 1978 Dissertation directed by: Dr. J. David Lockard and Dr. Dora K. Hayes

The purpose of this study was to determine if there is a link between adenyl cyclase activity and the diapause condition in the European corn borer, Ostrinia nubilalis. Insects inhabiting those latitudes where cold and warm seasons alternate with one another have evolved mechanisms which allow them to remain dormant (in a state of diapause) during the winter months of the year. Photoperiod, as well as temperature and humidity, has been shown to control the onset. maintenance and termination of insect diapause. In recent years, evidence supporting a role for the cyclic AMP system, including adenyl cyclase, as well as a role for one or more biogenic amines in the pathway between light reception and the neuroendocrine regulation of the insect life cycle and in the multitude of neuroendocrine pathways controlling insect growth and metamorphosis has been accumulating. In light of this evidence, it was decided to investigate the effects of two light regimens. short day (diapausing-inducing) and long day (pupationinducing), on adenyl cyclase activity of various stages

of fifth instar European corn borer larval heads, and to determine the effects of the biogenic amine neurotransmitters, norepinephrine, octopamine, and dopamine on this activity. Adenyl cyclase activity was measured by a modification of the method of Krishna, et al., (1968). A summary of the results follows.

In head extracts of fifth instar European corn borer larvae reared under both long day and short day photoperiodic regimens, adenyl cyclase activity in the presence of sodium fluoride increased as the larvae progressed through early, middle and mature stages. In long day larval heads, activity decreased in late prepupae and reached a low in pharate pupae. In contrast, adenvl cyclase activity in short day larval heads peaked in early diapause and then returned to prediapause levels during late diapause. Norepinephrine significantly enhanced adenyl cyclase activity only in early diapause larval head extracts, while octopamine significantly enhanced adenyl cyclase activity in head extracts of late short day mature and early diapause larvae. Dopamine was ineffective as an activator. An analysis of the combined effect of neurotransmitter and developmental stage revealed that in general, a given neurotransmitter in combination with short day larval head extracts resulted in higher adenyl cyclase levels than that neurotransmitter in combination with long day head extracts.

Based on these results, it appears that the cyclic AMP system is in some way involved in the initiation, maintenance, and termination of diapause in the European corn borer. While the exact nature of this involvement is open to speculation, several possible mechanisms for cyclic AMP's participation in the control of larval diapause are presented.

Suggestions for future research center around the determination of the circadian variations in adenyl cyclase levels in brains and corpora allata-cardiaca complexes in insects exposed to long day and short day photoperiodic regimens, and the determination of the effects of appropriate neurotransmitters, hormones, neurohormones and other peptides on this activity. Since the cyclic AMP system in diapausing European corn borers is functioning at a relatively high level, this system might provide a vulnerable point for attack against European corn borer infestation. Further, the assay used to measure adenyl cyclase activity may be useful in determining the effects of potential insect regulators on this system.

PREFACE

For any science course to meet the needs and interests of the student, it should make him aware of the following:

Science is a constantly changing body of knowledge.

2. Experimentation plays a vital role in the realm of science.

3. Careful observation and an eye for the unexpected are imperative in furthering scientific knowledge.

4. The scientific method is the backbone of scientific research.

5. Current science is based on past accomplishments.

6. One of the major goals of science is to benefit mankind.

The science educator, be he the science supervisor, curriculum specialist, or classroom teacher, must be equipped to insure that the appropriate learning experiences for meeting these objectives are provided. The study described herein has not only enabled the researcher to become more familiar with a vital area of biological research, namely that of the biochemistry of insect growth and metamorphosis, but also to experience the true nature of biological research as stated above.

The methodology and results of current research endeavors often serve as models for the student laboratory

ii

exercises of the next decade. In the field of biology, the study of biological rhythms has received attention in undergraduate as well as in graduate classes (Halberg et al. 1972; Koukkari et al. 1974). Investigation of the nature and control of insect biological rhythms is, then, a very appropriate area of investigation for the science educator.

An added contribution to the field of science education lies in the significance of the results of this study in gaining a better understanding of the role of neurohormones and neurotransmitters in learning and memory. The literature is filled with reports of links between neurohormones, neurotransmitters and many types of behavioral modification (de Wied and Gispen 1977). And as Hyden (1969) points out, it is likely that the biochemical mechanisms which operate to control innate or instinctive behaviors associated with learning and memory. Therefore, the knowledge gained from the research described herein may prove to be very useful in furthering our understanding of the neuroendocrine control of learning.

iii

ACKNOWLEDGMENTS

To Drs. J. David Lockard and Dora K. Hayes, my major advisors, I wish to express my gratitude for their direction and support. From them, I received enough guidance and supervision to insure success, but not so much as to stifle creativity and dampen the independent spirit.

To Dr. Emmett Wright, many thanks for your direction and suggestions in the development of my research proposal.

To Dr. William Schafer, my appreciation for your counsel in designing and conducting the statistical analyses used in these investigations.

To Dr. John Davidson, my thanks for your support as a member of my dissertation committee.

This study was conducted under the auspices of Dr. Dora K. Hayes of the Chemical and Biophysical Control Laboratory, USDA, Beltsville, Maryland. Research facilities and support for the study were provided by the Department of Agriculture's Science and Education Administration, Washington, D. C. The European corn borers used in these investigations were provided by the USDA Corn Insect Laboratory, Ankeny, Iowa.

Finally, I wish to express my deepest gratitude to my husband, Hy, my son, Jay, my mother, Ida, my father, Ben, and to Virginia, without whose devotion, cooperation, and understanding this dissertation would not have been possible.

iv

TABLE OF CONTENTS

																	Page
PREFACE		0 0			• •		6	0	0								ii
ACKNOWL	EDGMENTS	5 8			* 0	•	٠	•		•	• •		×				iv
LIST OF	TABLES.	• •		•	• •		0		•	6 (•	9	٠		•		vii
LIST OF	FIGURES	• •	•			•	•		•					•		٠	ix
CHAPTER																	
I.	INTRODU	JCTI	ON	•		8	8	8	0	• •	•			•			1
II.	Purpose Stateme Stateme Definit Limitat Procedu Signifi Organiz REVIEW The Eur Chara The Dia The Dia The Cor Facto The "Pr	e of ent ion tion tion are. can tro of of ors. noto	th of of s c c on THE an ris s c e on THE an ris s e l c	ne f the the of f of f of con tic Sta	Stud e Pr e Hy Ferr the the the ITE ITE CS. ate Diap	ly rok ypc St St Sor Sor Cl	ole oth tud Stu Stu FUR rer see	em les y dy dy dy E b	es • • • • • • • • • • • • • • • • • •	ner Phy		·	• • • • • • • • • • • • • • • • • • •	0 0 0 0 0 0 0 0		• • • • • • •	11 12 13 15 17 19 21 22 24 27 29
	Mecha Mecha	nis	or m. ndo	ver	ne			no ro	ůr.	of	.SS The	•	0+		•		34
	Metan A Role	for	hos	is	and ic A	1 I AMF)ev	el	opi	men	t.	•	•	•	8	•	39
	Growt	ch a:	nd	Met	tamo	orp	ho	si	S	6 a				0	•		52
III.	MATERIA	LS	ANE	MI	ETHC	DDS		•	Ð					•	e	•	66
	Maintai Identif The Ade Calcula Statist	nin Tica enyl tion tica	g C tio Cy n o l A	orr cla f A nal	n Bo of H ase der ysi	Fif As nyl	r 'th sa C	Ia I y yc	rva ns las	ae. tar se	St Act	tag tiv	es it	• • У	•	0 0 9 0	67 67 75 76
IV.	RESULTS	. 0				•	•	0		• •		•		•			78

Page

CHAPTER

	Comparison of Adenyl Cyclase Activity in Head Extracts of Various-Aged Fifth Instar European Corn Borer Larvae Exposed to Photoperiods of Either LD 16:8 (LD) or LD 10:14 (SD)	70
	Effects of Norepinephrine, Dopamine, and Octopamine on Adenyl Cyclase Activity in Heads of Mature LD, Late Prepupae, Pharate Pupae, Late Mature SD, Farly Diapause, Late Diapause	78
	and Refrigerated Diapause Fifth	
	Instar Larvae in the Presence of Sodium Fluoride	86
	Activity in Head Extracts of LD Mature, Late Prepupal, Pharate Pupal, Early SD Mature, Late SD Mature, Early Diapause, Late Diapause, and Refrigerated Diapause Fifth Instar Larwage in the Absence of Sodium	
	Fluoride.	89
	Summary of Results	95
V.	DISCUSSION OF RESULTS AND RECOMMENDATIONS	97
APPENDIX-	-SUPPLEMENTARY TABLES	108
LIST OF A	BBREVIATIONS	118
BIBLIOGRA	PHY	120

LIST OF TABLES

Table

1.	One Way ANOVA Summary Table for the Determination of Significant Differences in Adenyl Cyclase Activity (Expressed as Moles/Head/30 Minutes) Among Stages of LD 16:8 and LD 10:14 Fifth Instar Larval Heads.	80
2.	One Way ANOVA Summary Table for the Determination of Significant Differences in Protein Concentration Among Stages of LD 16:8 and LD 10:14 Fifth Instar Larval Heads	83
3.	One Way ANOVA Summary Table for the Determination of Significant Differences in Adenyl Cyclase Activity (Expressed as Moles/Head/Milligram Protein) Among Stages of LD 16:8 and LD 10:14 Fifth Instar Larval Heads.	85
4.	Adenyl Cyclase Activity of Head Extracts of Mature LD 16:8 Fifth Instar Larvae in the Absence of Theophylline and in the Absence of Theophylline and the Presence of Phosphodiesterase.	86
5.	A Summary of the Effects of Norepinephrine, Dopamine, and Octopamine on Adenyl Cyclase Activity of Seven Stages of LD 16:8 and LD 10:14 Fifth Instar Larval Heads in the Presence of Sodium Fluoride	88
6.	Two Way ANOVA Summary Table for the Determination of the Presence of Main Effects and Interaction in the Analysis of the Effects of Norepinephrine, Dopamine, and Octopamine on Adenyl Cyclase Activity of Seven Stages of LD 16:8 and LD 10:14 Fifth Instar Larval Heads in the Presence of Sodium Fluoride	89
7.	A Summary of the Effects of Norepinephrine, Dopamine, and Octopamine on Adenyl Cyclase Activity of Eight Stages of LD 16:8 and LD 10:14 Fifth Instar Iarval Heads in the Absence of Sodium Fluoride	90

Table

8.	Two Way ANOVA Summary Table for the Determination of the Presence of Main Effects and Interaction in the Analysis of the Effects of Norepinephrine, Dopamine, and Octopamine on Adenyl Cyclase Activity of Eight Stages of LD 16:8 and LD 10:14 Fifth Instar Larval Heads in the Absence of Sodium Fluoride
9.	Adenyl Cyclase Activity (Expressed as Moles/Head/30 Minutes) in Head Extracts of LD 16:8 and LD 10:14 Fifth Instar Larval Heads
10.	Milligrams of Protein Per Head of LD 16:8 and LD 10:14 Fifth Instar European Corn Borer Larvae
11.	Adenyl Cyclase Activity (Expressed as Moles/Milligram Protein/30 Minutes) in LD 16:8 and LD 10:14 Fifth Instar Iarval Heads
12.	Effects of Norepinephrine, Dopamine, and Octopamine on Adenyl Cyclase Activity of Seven Stages of LD 16:8 and LD 10:14 Fifth Instar Larval Heads in the Presence of Sodium Fluoride
13.	Effects of Norepinephrine, Dopamine, and Octopamine on Adenyl Cyclase Activity of Eight Stages of LD 16:8 and LD 10:14 Fifth Instar Larval Heads in the Absence of Sodium Fluoride

LIST OF FIGURES

Figure		Page
1.	Diagrammatic Representation of the Anatomical Interrelationships of the Various Structures of the Insect Neuroendocrine System.	4
2.	Diagrammatic Representation of the Endocrine System's Control of Insect Growth and Metamorphosis	6
3.	Model for the Production of <i><</i> -Ecdysone	8
4.	Adenyl Cyclase Involvement in the Two- Messenger System of Hormone Action	9
5.	European Corn Borer AbundanceFall, 1974	23
6.	The Larval, Late Prepupal, Pharate Pupal, Pupal, and Adult Stages of the European Corn Borer	26
7.	The Effect of Photophase Duration on Diapause Incidence Among European Corn Borer Larvae Reared Under a 24-Hour Photoperiod and a Temperature of 23° C	31
8.	Lateral View of the Brain and Neuro- endocrine Organs of the European Corn Borer	47
9.	The Neurosecretory Cells of the Larval Brain of the European Corn Borer	48
10.	The Formation and Breakdown of Cyclic AMP	55
11.	The Early, Middle, and Mature Fifth Instar Stages of the European Corn Borer	69
12.	Adenyl Cyclase Activity in Head Extracts of LD 16:8 and LD 10:14 Fifth Instar European Corn Borer Larvae Expressed as Moles/Head/30 Minutes.	79
13.	Milligrams of Protein in Heads of LD 16:8 and LD 10:14 Fifth Instar European Corn Borer Larvae	81

Figure

Page

CHAPTER I

INTRODUCTION

The European corn borer, Ostrinia nubilalis (Hübner), is an insect pest that attacks corn and other crops. The corn borer undergoes complete metamorphosis; eggs hatch to become voraciously hungry larvae. During the larval stage the corn borer generally undergoes four molts, thus passing through five larval stages or instars. Fifteen to thirty days after hatching, depending on environmental conditions, larvae undergo metamorphosis and change into pupae. It is during the pupal stage that the larvae become transformed into adults. Adults emerge from the pupal case after one to two weeks.

Insects inhabiting latitudes where cold and warm seasons alternate with one another have evolved mechanisms which allow them to remain active during the spring, summer, and fall, and dormant during the winter months of the year. Thus, insects are able to cope with rather unfavorable winter conditions and be active when the probability for their survival is greatest. Winter diapause or "hibernation" may occur in any stage of insect development, but each insect species will always diapause at a particular stage-either egg, larva, pupa or adult. It is the fifth instar corn borer larva that will enter diapause during the fall. Diapause is broken in the spring and pupal formation ensues. This is brought about by the increase in the number of hours of daylight and the rise in the ambient temperature

which induce the release of prothoracicotropic hormone (PTTH) from the brain. PTTH induces a chain of events which results in pupation. It is believed that a similar phenomenon must occur for the breaking of insect diapause. All the events which occur between light reception and neuroendocrine activation are not yet known. However, there is evidence to show that cyclic AMP and one or more biogenic amines may be involved.

According to Minis (1965), Kogure (1933) was the first to demonstrate that insect diapause was under the control of changes in day length, or photoperiod. Thirteen to fourteen hours or less of daylight, depending on the species, accompanied by temperatures of 25 degrees centigrade or below have been found to induce diapause in lepidopterous insects (moths and butterflies) (Lukefahr 1961: Adkisson 1965; Williams 1969). Since change in day length appears to be the environmental stimulus which triggers the onset of diapause as well as its termination (Adkisson 1965), insects must possess a photoreceptor for detecting light, a "biological clock" which keeps a record of the hours of illumination and/or darkness, and a mechanism for activating the endocrine system which controls insect growth and development accordingly. While a great deal of research has been directed towards elucidating the mechanisms involved in the initiation and termination of diapause, there are still a great many questions that remain to be answered.

These include: 1. the identity of the photoreceptor, 2. the mechanism by which the biological clock records hours of illumination, and 3. the mechanism by which the endocrine system is coupled to the "biological clock." This dissertation will concern itself with the third of these questions.

Figure 1 (from Herman and Gilbert (1966)) is a diagrammatic representation of the neuroendocrine structures which control insect diapause and metamorphosis. Williams (1969) purports that it is the brain of the oak silkworm, Antheraea pernyi, that is the receptor for photoperiodic signals. The brain interprets these signals, and when appropriate, produces PTTH which is stored in and released into the haemolymph (blood) from the corpora cardiaca (Highnam 1958; Scharrer 1952). This hormone in turn stimulates the prothoracic glands to produce molting hormone (Wigglesworth 1964; Vedeckis et al. 1974). The actual product of the prothoracic glands has been shown to be ~-ecdysone (King et al. 1974; Chino et al. 1974). ~-ecdysone is converted to β -ecdysone, the true molting hormone, by peripheral body tissues (Bollenbacher et al. 1975). It is $\beta_{-ecdysone}$, then, that is believed to elicit those biochemical and morphological events commonly known as molting. In the presence of high titres of juvenile hormone, a product of the brain's corpora allata, a larval molt will occur. When titres of juvenile hormone are low, the larva will

Fig. 1. Diagrammatic representation of the anatomical interrelationships of the various structures of the insect neuroendocrine system (not to scale). B, brain; SEG, subesophageal ganglion; CA, corpus allatum; CC, corpus cardiacum; PTG, prothoracic ganglion; MTG, mesothoracic ganglion; MeTG, metathoracic ganglion; PG, prothoracic gland. (From Herman and Gilbert 1966).

4a



metamorphosize into a pupa. In the absence of juvenile hormone, a pupal-adult molt will occur. Figure 2 (from Schneiderman and Gilbert (1959)) is a diagrammatic representation of how the insect endocrine system is involved in the growth and differentiation of lepidopterous insects.

Photoperiod appears to control insect diapause by regulating the synthesis, release and/or transport of PTTH (Williams and Adkisson 1964), for larvae will enter diapause when ecdysone production has been shut down. Recently, considerable evidence has accumulated linking an intermediate titre of juvenile hormone with this shutdown (Chippendale 1977).

Neuroendocrine control of insect growth and differentiation is further complicated by the changing pattern of responsiveness of the target tissues to hormone action (Bodenstein 1957) at specific times in the developmental schedule. For example, Truman and Riddiford (1974) have shown that tobacco hornworm, <u>Manduca sexta</u>, larvae will only enter the wandering stage during hours of darkness, 2400-1200 (A.Z.T.)¹, even though these larvae have already become competent to release the hormone (ecdysone) which initiates the onset of wandering.

¹A.Z.T. stands for arbitrary "zeitgeber" time where 'zeitgeber' refers to an environmental factor, in this case the onset of darkness, which is capable of synchronizing a circadian periodicity.

Fig. 2. Diagrammatic representation of the endocrine system's control of insect growth and metamorphosis. PTTH, prothoracicotropic hormone; MH, molting hormone; JH, juvenile hormone. (From Schneiderman and Gilbert 1964).



ADULT

ba.

Recently, evidence has been presented to show the involvement of the cyclic AMP system of thetobacco hornworm in PTTH's stimulation of *d*-ecdysone production. Figure 3 (from Vedeckis et al. (1974)) illustrates the currently accepted model for the production of *«-ecdvsone.* The synthesis of adenosine 3',5'-cyclic monophosphate (cyclic AMP) is directed by the enzyme, adenyl cyclase; its breakdown is directed by the enzyme, cyclic AMP phosphodiesterase. Rasenick et al. (1976) have associated an increase of cyclic AMP with the secretion of PTTH by the brain of the Cecropia silkmoth, Hyalophora cecropia. The role of cyclic AMP in these two instances and in general. is believed to be that of a second messenger (Sutherland et al. 1965). Thus some transmitter or hormone stimulates the production of adenyl cyclase which in turn results in increased levels of cyclic AMP. Cyclic AMP, then, acting as a second messenger, stimulates enzyme production which initiates the appropriate biological response associated with the original transmitter. (See Figure 4).

In vertebrates, adenyl cyclase has been linked to the light-regulated circadian rhythm of serotonin and N-acetyltransferase activity (Klein and Berg 1970; Romero and Axelrod 1974). Circadian rhythms are defined as those endogenous biological rhythms whose period is an approximation to the period of the earth's rotation (Halberg 1959). Perhaps adenyl cyclase is similarly involved in

Fig. 3. Model for the production of *L*-ecdysone. (From Vedeckis et al. 1974.)



Sa

Fig. 4. The two-messenger system of hormone action involving adenyl cyclase. (From Sutherland et al. 1965.)

9a



EFFECTOR CELL

RESPONSE

ACTION OF SEVERAL HORMONES OR NEUROHUMORS

the light-regulated mechanism which controls insect diapause. Preliminary results of Rasenick et al. (1978) support the presence of diurnal rhythms of cyclic AMP in the brains of <u>Manduca sexta</u> and of Cecropia. If as Weiss and Strada (1972) believe, environmental lighting is one of the prime regulators of the cyclic AMP system, these rhythms may be monitoring photoperiodic information.

A stimulatory effect of norepinephrine on adenyl cyclase activity has been demonstrated in vertebrate pineal glands (Weiss and Costa 1968a). Of great significance is the finding that time of day influences the relative sensitivity of pineal gland adenyl cyclase activity to norepinephrine. A similar stimulatory effect has been found in adult cockroach brain extracts (Rojakovick and March 1972). Effects of photoperiod on this activity have not yet been investigated. The biogenic amines, octopamine, dopamine, and serotonin have also been found to stimulate adenyl cyclase activity in nervous tissue of the American cockroach (Nathanson and Greengard 1973).

In summary, insect diapause is controlled by the photoperiodic regimen to which the insect is subjected. The brain houses the photoperiodic control mechanism. Adenyl cyclase activity has been shown to play an important role in insect metamorphosis and possibly the control of circadian rhythms. Neurotransmitters such as norepinephrine, octopamine, and dopamine appear to stimulate

adenyl cyclase activity in certain insects. In light of these findings, it was decided to investigate the effects of two light regimens, one pupa-inducing, the other diapause-inducing, on adenyl cyclase activity of various stages of fifth instar corn borer larval heads, and to determine the effects of the neurotransmitters, norepinephrine, octopamine and dopamine on this activity.

PURPOSE OF THE STUDY

The purpose of this study was to determine if there is a link between adenyl cyclase activity and the diapause condition in the European corn borer, <u>Ostrinia nubilalis</u>.

STATEMENT OF THE PROBLEM

The problem was to determine if adenyl cyclase is involved in the onset, potentiation, and/or termination of diapause in the European corn borer, <u>Ostrinia nubilalis</u>, by measuring adenyl cyclase activity of various-aged diapausing and non-diapausing fifth instar larval heads and by determining the effects of norepinephrine, octopamine, and dopamine on this activity.

STATEMENT OF THE HYPOTHESES

1. There is no difference in adenyl cyclase activity among early and late diapausing, and various-aged fifth instar LD 16:8 (LD) and non-diapausing LD 10:14 (SD) larval heads of the European corn borer.¹

2. There is no effect of norepinephrine, dopamine or octopamine on adenyl cyclase activity in selected stages² of LD and SD fifth instar larvae.

3. There is no difference in the effects of norepinephrine, dopamine, and octopamine on adenyl cyclase activity within any one of the fifth instar stages selected.

4. There is no difference in the effect of any one of the neurotransmitters on adenyl cyclase activity among the fifth instar stages selected.

²The selection of stages was based upon the results of those experiments designed to test hypothesis one.

¹For LD 16:8, adenyl cyclase activity in head extracts of early, middle, mature, early prepupal, late prepupal, and pharate pupal fifth instar larvae was determined. For LD 10:14, adenyl cyclase activity in head extracts of early, middle, early mature, late mature, early diapause, late diapause, and refrigerated diapause fifth instar larvae was determined.

DEFINITION OF TERMS

1. CIRCADIAN RHYTHM: that internal, biological rhythm whose period is an approximation of the period of the earth's rotation.

2. COMPLETE METAMORPHOSIS: metamorphosis characteristic of holometabolous insects. The egg hatches into a larvae (often called a caterpillar or maggot) which undergoes several molts before changing into a pupa. It is during the pupal rather than the larval stage that there is considerable change of form in the direction of the adult. Thus, after a spectacular metamorphosis, the adult emerges from the pupal case.

3. DIAPAUSE: a period of arrested development similar to hibernation.

4. EARLY DIAPAUSE LARVA: a mature fifth instar larva (LD 10:14, maintained at a temperature of $24^{\circ}C \pm 1.5^{\circ}C$) that is between 40 and 43 days of age.

5. EARLY FIFTH INSTAR LARVA: a fifth instar larva whose body diameter at the midpoint is between 1.0 and 1.5 mm.

6. EARLY MATURE FIFTH INSTAR LARVA: a two to three week old mature fifth instar larva.

7. EARLY PREPUPA: a prepupa which has not enclosed itself in silk.

8. INSTAR: the stage of an insect between successive larval molts; the first instar being the stage between hatching and the first molt.

9. LATE DIAPAUSE LARVA: a mature fifth instar larva (LD 10:14, maintained at a temperature of 24°C ± 1.5°C) which is between 45 and 55 days of age.

10. LATE PREPUPA: a prepupa which has enclosed itself in silk.

11. LD 16:8: a light regimen characterized by sixteen hours of light and eight hours of dark.

12. LD 10:14: a light regimen characterized by ten hours of light and fourteen hours of dark.

13. MATURE FIFTH INSTAR LARVA: a fifth instar larva whose body diameter at the midpoint is 3.0 \pm ^{0.3} mm.

14. MEASURE OF ADENYL CYCLASE ACTIVITY: amount of ATP converted to cyclic AMP as determined by a modification of the method of Krishna et al. (1968).

15. MIDDLE FIFTH INSTAR IARVA: a fifth instar larva whose body diameter at the midpoint is 2.0 \pm 0.3 mm.

16. METAMORPHOSIS: a change in insect form during development.

17. MOLT: a process of shedding the exoskeleton (external skeleton); ecdysis.

18. NEUROHORMONE: a regulatory substance produced by nervous tissue at extrasynaptic sites.

19. NEUROTRANSMITTER: a substance produced by nerve cells that is responsible for synaptic transmission.

20. PHARATE PUPA: a mature fifth instar larva (LD 16:8, maintained at a temperature of $24^{\circ}C \pm 1.5^{\circ}C$) characterized by a relative quiescence, light body color, reduced body turgidity and sigmoid shape.

21. PHOTOPERIOD: the number of hours of light and dark in a given period of time, typically 24 hours.

22. PHOTOPHASE: the light portion of the photoperiod.

23. PREPUPA: a mature fifth instar larva (LD 16:8, maintained at a temperature of $24^{\circ}C \pm 1.5^{\circ}C$) whose gut contents are white in color.

24. REFRIGERATED DIAPAUSE LARVA: a mature fifth instar LD 10:14 larva that has been refrigerated under a light regimen of LD 10:14 and a temperature of $4^{\circ}C \pm 1.5^{\circ}C$ for four to six weeks after having entered diapause.

25. SCOTOPHASE: the dark portion of the photoperiod.

LIMITATIONS OF THE STUDY

1. The insects used for the study are laboratoryreared. Although strains are always rejuvenated by periodic matings with field insects, differences may exist in adenyl cyclase activity of laboratory-reared and field strains.

2. Conditions for the maintenance of insect cultures are designed to simulate those of the field. Such simulation is never perfect. 3. Results cannot be generalized to all lepidopterous insects, much less to all insects. As Judy (1974) explains, it is very likely that differences exist among insect species in terms of the mechanisms which regulate the timing of a molt.

4. Differences in developmental rates exist even among insects exposed to identical environmental conditions. Thus, insects of the same age were actually at slightly different stages of development. This increased the variability of the results.

5. Approximately five percent of LD 16:8 larvae did not pupate while approximately ten percent of LD 10:14 did. This resulted in a slight decrease in the reliability of the results for non-diapausing insects.

6. The differences which exist in adenyl cyclase levels among the various groups of larval heads cannot be attributed to any particular neuroendocrine structure.

7. Effects of norepinephrine, dopamine, and octopamine are stated in terms of specific concentrations used. It is possible that other concentrations might have had different effects.

8. All larvae were decapitated between the hours of 11 A.M. and 1 P.M. Since it is plausible to expect circadian variations in adenyl cyclase levels of head homogenates, values for adenyl cyclase activity would probably differ had the insects been decapitated at a different time during the day.

PROCEDURE

Second instar larvae received from the European Corn Borer Laboratory (USDA) in Ankeny, Iowa were randomly divided into two groups. One group was placed in a Biological Oxygen Demand Box (BOD) set to maintain a light regimen of sixteen hours of light and eight hours of dark (LD 16:8) and a temperature of 24 ± 1.5 degrees centigrade. The second group of larvae was placed in a similar box set to maintain a light regimen of ten hours of light and fourteen hours of dark (ID 10:14). These photoperiodic regimens were found to be ideal for inducing pupation and diapause respectively in LD and SD Ankeny corn borers. Fifth instar larvae were identified by their head capsule width (1.70-2.10 mm). Early, middle. and mature fifth instar larvae were identified by their body diameter, prepupae by the white color of their gut contents. and pharate pupae by their relative quiescence, light body color, reduced body turgidity and forward curvature of the head into a somewhat tucked position. LD 10:14 fifth instar larvae were assumed to be in diapause if they had not undergone pupation by the time they had reached forty days of age.

Adenyl cyclase activity of larval heads was measured by a modification of the method of Krishna et al. (1968). This assay is designed to determine the amount of radioactive cyclic AMP produced from radioactive adenosine triphosphate (ATP) due to the action of the enzyme adenyl cyclase. Breakdown of the cyclic AMP produced was prevented by the addition of a cyclic AMP phosphodiesterase inhibitor,

theophylline. The radioactive cyclic AMP was purified by chromatography on Dowex 50 H⁺ ion exchange columns followed by treatment with zinc sulfate and barium hydroxide. The amount of radioactive cyclic AMP produced as a result of adenyl cyclase activity was measured in a Beckman liquid scintillation counter.

A flow chart of the research design follows:

Second instar larvae randomly divided into two groups and placed in appropriate Biological Oxygen Demand Boxes. Time allowed for maturation. Random selection of: LD 16:8 fifth instar larvae (early (ELD), middle 1. (MILD), mature (MLD), early prepupae (EPP), late prepupae (LPP), pharate pupae (PHP)) 2. Non-diapausing LD 10:14 fifth instar larvae (early (ESD), middle (MISD), early mature (ESDM), late mature (LSDM)) Diapausing larvae (early (EDIAP), late (LDIAP), 3. refrigerated (RDIAP).

Determination of adenyl cyclase activity in extracts of larval heads of each of the thirteen groups mentioned above.
In vitro determination of the effects of norepinephrine, dopamine, and octopamine on adenyl cyclase activity of head extracts of selected fifth instar stages.

Statistical analysis of the data.

To test hypothesis one, a thirteen group one way ANOVA ($\measuredangle = 0.05$) was performed. The Student Newman-Keuls procedure, as described in Winer (1962), was used as a post hoc test. To test hypotheses two, three, and four, a two way ANOVA (stages x neurotransmitters) was performed ($\measuredangle = 0.05$). The Student Newman-Keuls procedure was also used as a post hoc test.

SIGNIFICANCE OF THE STUDY

The importance of the insect brain as an endocrine organ responsible for regulating insect molting and metamorphosis as well as its light-sensing role in the initiation and termination of diapause is well-documented. However, the

exact mechanisms by which it accomplishes these tasks. especially its control over diapause is still a mysterv. The results of this study have provided additional information concerning the biophysical and biochemical nature of insect diapause. By subjecting larvae to light regimens which were either pupa-inducing or diapause-inducing and measuring adenyl cyclase levels, the researcher was able to determine that there is a link between adenyl cyclase activity and the diapause condition in the European corn borer. By determining the effects of selected neurotransmitters on adenyl cyclase activity, the researcher was able to ascertain the relative sensitivity of adenyl cyclase receptors to these neurotransmitters during the various stages of the fifth instar studied. These findings have helped to elucidate the mechanism by which the neuroendocrine system controls insect diapause and should be useful in the consideration and design of future methods of insect Insects in diapause are typically more resistant control. to insect control procedures than those which are not in diapause. Diapausing corn borers are well concealed and are protected by a silken case. In addition, their overall metabolic rate is considerably reduced. In contrast, the cyclic AMP system in diapausing borers appears to be functioning at a relatively high level. Therefore, future control efforts might be directed at interfering with this

particular system. The assay developed to measure adenyl cyclase activity could be employed to detect the effects of potential insect regulators on this system.

ORGANIZATION OF THE STUDY

The remainder of this study is divided into four chapters. Chapter two presents a review of the literature. It includes a discussion of:

- 1. the life cycle of the European corn borer
- 2. characteristics of the diapause state
- 3. the control of diapause by physical factors

4. the roles of the various neuroendocrine structures (brain, corpora cardiaca, corpora allata, and prothoracic glands) in controlling insect metamorphosis and diapause

5. the involvement of the cyclic AMP system in the neuroendocrine control of insect metamorphosis and diapause. Chapter three describes the methodology used to rear and select insects, to assay for adenyl cyclase activity, and to statistically analyze the results. Chapter four presents the experimental results as well as an interpretation of these results. Chapter five analyzes and discusses the significance of the results, and relates them to the findings of other researchers. In addition, suggestions for future research are presented.

CHAPTER II

REVIEW OF THE LITERATURE

According to Shapiro and Pereverzev (1974), the European corn borer, <u>Ostrinia nubilalis</u> (Hübner), is one of the most widespread and destructive corn pests in the world. Ten to fifteen years ago, before the development of resistant hybrids, annual losses of corn in the United States due to European corn borer activity was estimated at 300 million dollars. In 1974 because of the use of resistant hybrids, this loss had been reduced to 140 million dollars. In the North Central United States, loss due to the larval activity of the corn borer has been estimated at between two and four percent per borer per plant. Figure 5 (from the USDA Coop. Econ. Insect Rep. 1975) shows the relative abundance of the corn borer in the North Central and Eastern United States as of 1974.

Ryder et al. (1969) have reported that the European corn borer was first discovered in the United States about 1917. This lepidopterous insect pest also occurs in other countries throughout the world including Egypt, Canada, the Soviet Union, Germany, Bulgaria, Japan, China, and the Phillipines (El-Minshawy et al. 1974).

The European corn borer attacks mainly maize (Zea mays) doing a great deal of damage to stalks, ears, and tassels. Damage done to food conducting vessels by early generation larvae results in plant weakness, reduction of

Fig. 5. European corn borer abundance, Fall 1974. (From the USDA Coop. Econ. Insect Rep. 1975.)



ear size and weight, and reduction of grain number and grain weight. Late generation larvae cause loss of ears due to ear dropping (Chiarappa 1971).

THE EUROPEAN CORN BORER--GENERAL CHARACTERISTICS

The European corn borer is a member of the insect order Lepidoptera (moths and butterflies); family, Pyralidae; subfamily, pyraustinae. It is a yellowish-brown moth whose wingspan is just over one inch. As an adult it has scales which cover its wings and most of its body and legs, sucking mouthparts, large compound eyes composed of many facets, and two ocelli (simple eyes).

The corn borer is classified as an holometabolous insect, one which undergoes complete metamorphosis. Its life cycle includes the stages of egg, larva, pupa, and adult. In the spring eggs are laid in small clusters containing approximately 20-25 individuals. Eggs hatch in three to five days. The larva is eruciform having a welldeveloped head and a cylindrical body consisting of three thoracic and ten abdominal segments. The head bears two short antennae, chewing mouthparts, and 12 ocelli (six on each side just above the mandibles). Well-developed silk glands open on the lower lip. One pair of legs is present

on each thoracic segment; and one pair of prolegs is found on abdominal segments three through six and segment ten. During a period of between 15 and 30 days (depending on environmental conditions) larvae generally undergo four ecdvses (molts) before metamorphosing into pupae. Thus. there are typically five larval stages each known as an instar. Particular instars are identified by head capsule width (Vinal and Caffrey 1919; Gelman and Hayes 1978). The fifth larval instar may reach a length of up to 25 mm. Towards the end of the fifth instar, larvae cease to feed and are then designated prepupae. Prepupae enclose themselves in silken cases and enter the pharate pupa stage characterized by a sigmoid shape, head bent forward into a somewhat tucked position. In less than 24 hours the cuticle is shed and the pupal stage begins. Pupae are obtect (appendages more or less glued to the body), brown in color, and generally 10 to 12 mm in length. It is during the pupal stage, which lasts for about one week, that larvae metamorphose into adult moths. Figure 6 depicts the larval, prepupal, pharate pupal, pupal and adult stages in the corn borer life cycle.

Fig. 6. The larval, late prepupal, pharate pupal, pupal, and adult stages of the European corn borer. Silken cases which enclose the late prepupa and pharate pupa have been split open so that the animal can be seen.

26a B A Concession of 1 CM 1 D С E

THE DIAPAUSE STATE

Insects inhabiting those latitudes where cold and warm seasons alternate with one another have evolved mechanisms which allow them to endure the low temperatures of the late fall and winter seasons. They enter a physiological state known as diapause, which is characterized by a lack of growth and morphogenesis, and which is due to profound changes in insect metabolism. The ability of an insect to enter diapause is determined genetically and may be under polygenic control (King 1974; Hong and Platt 1975). Beck (1962a) originally defined diapause as "a state of arrested development in which the arrest is enforced by a physiological mechanism rather than by unfavorable environmental conditions." However, in 1964, Beck and Alexander showed that development does occur during diapause, but at a very slow rate. Therefore, a more appropriate definition for diapause is a "state of suppressed development" (Beck 1968).

Diapause may occur in any stage of insect development (egg, larva, nymph, pupa, or adult); but in any given species, it will always occur in the same stage. Insect species may be classified according to the type of diapause they exhibit. Univoltine species will diapause in each generation at some stage during their life cycle and are therefore said to have obligatory diapause. They only produce one generation per

year. Polyvoltine species generally produce more than one generation per year. Those with facultative diapause may or may not diapause depending on environmental conditions, while polyvoltine nondiapausing insects continue their development throughout the year.

The European corn borer is a polyvoltine species with a facultative diapause. In spring and early summer, the corn borer completes its normal life cycle; but, in late summer and early fall, fifth instar larvae will enter diapause and overwinter in this state. Diapausing larvae exhibit reduced movement, reduced oxygen consumption and water content, increased fat reserves, and an overall lowering of metabolism (RNA and DNA synthesis, protein synthesis, and enzyme levels) (Beck and Hanec 1960; Schechter et al. 1971; Brown and Chippendale 1977). It is known that during diapause the corn borer can withstand prolonged periods of low temperatures (down to -30° to -40°C), temperatures which they could not tolerate in the nondiapause state (Maslennikova 1973). Anatomically, diapausing male larvae can be distinguished from nondiapausing larvae on the basis of gonadal development. As early as the fourth instar, male larvae programmed to enter a state of diapause showed underdeveloped rudimentary testes (Parker and Thompson 1927).

THE CONTROL OF DIAPAUSE BY PHYSICAL FACTORS

Since seasonal changes are accompanied by changes in the number of daylight hours per 24-hour day, and since these changes are exactly the same from year to year, insects have evolved the ability to use photoperiod (the number of hours of light and dark in a given period of time, typically 24 hours) to regulate their life cycles. Thus, insects do not enter diapause as a result of the relatively sudden onset of unfavorable conditions, but rather as a result of the changes in day length and temperature to which prediapausing stages have been exposed. As the days become shorter in late summer, the corn borer uses this signal to physiologically prepare itself for Thus, larval diapause in this species, as in many winter. others, has been shown to be in response to short days and low temperatures (Mutchmor and Beckel 1959; Beck and Hanec 1960).

Beck and Hanec (1960) have demonstrated that at a temperature of 23° C, the critical photophase (span of light) for the Northern Wisconsin European corn borer is between 15 and 15.5 hours of light for a 24 hour cycle.¹ Thus, corn borers exposed to a photophase of 15.5 hours accompanied by a scotophase (period of darkness) of 8.5

¹Different strains of European corn borers have evolved different critical daylengths in response to the seasonal conditions of temperature accumulation and daylengths in the specific locality in which they live (Beck 1963). These critical photoperiods vary between 14 and 15.5 hours.

hours exhibited 18% diapause, while those exposed to a photophase of 15 hours accompanied by a scotophase of 9 hours exhibited 83% diapause. Figure 7 from Beck (1962a) illustrates the effect of photoperiod on the incidence of diapause among European corn borer larvae reared at a temperature of 30° C. As can be seen, at this temperature, diapause is induced by a relatively narrow range of photoperiods, those having a photophase of between 10 and 14 hours. Similar curves have been found for other insect species (Adkisson 1965; Beck 1968; Lees 1968).

That corn borer larvae are exceptionally sensitive to photoperiod is shown by the finding that diapause can be prevented in up to 70% of corn borer larvae reared in diapause-inducing photoperiods by transferring them to a non-diapause-inducing photoperiod as late in development as the fifth instar (Beck and Hanec 1960). Then too, an interruption in the scotophase of as little as 0.5 hours of light greatly reduced diapause incidence (Beck 1962a).

Beck (1962a) also found that the duration of the scotophase is actually more important than that of the photophase in inducing corn borer larvae to diapause. A scotophase of twelve hours was found to yield maximum diapause when combined with photophases of from five to eighteen hours. Even a nondiel (cycle other than 24 hours) photophase of up to 32 hours yielded significant diapause when used in conjunction with a 12 hour scotophase.

Fig. 7. The effect of photophase duration on diapause incidence among European corn borer larvae reared under a 24-hour photoperiod and a temperature of 23° C. (From Beck 1962a.)

3/2



Temperature, too, exerts an influence on the onset of diapause. While photoperiods having a 10.5 to 13.5 hour scotophase will induce diapause at all temperatures between 20° and 30°C, Beck (1962b) reported that under other photoperiodic regimens, the onset of diapause appeared to be inversely proportional to the rearing temperatures. This temperature sensitivity was only exhibited during the scotophase. He also demonstrated that thermoperiods involving steep temperature gradients could override photoperiod in the induction of diapause. Thus, whether under constant light or constant darkness (non-diapause-inducing photoperiods) corn borers maintained at 31°C for 11 hours and then 10°C for 11 hours (2 hours for transition) entered diapause.

Andrewartha (1952) originated the term "diapause development" to describe the physiological changes that must take place for the termination of diapause. The photoperiod and temperature range required for "diapause development" of the European corn borer is nearly the same as that for morphogenesis (McLeod and Beck 1963a). Thus, according to McLeod and Beck, diapausing larvae exposed to LD 16:8 (16 hours of light, 8 hours of dark) pupated after 30 days while those exposed to LD 8:16 pupated after 70 days.

Conditions necessary for the termination of diapause have been summarized by Beck (1963). He noted that while many species require chilling for several weeks in order to break diapause, the corn borer did not. In addition, according to Bowen and Skopik (1976), it is the duration of the scotophase rather than the duration of the photophase which appears to be important in terminating diapause. It should be noted that in the field, diapausing borers who have completed diapause development by early winter will not pupate until spring in response to increased moisture and rising temperatures. In the laboratory, the need for sufficient moisture is shown by the inability of long-day photoperiods to terminate diapause in partially dehydrated borers (Beck 1967); the need for increasing temperature is shown by the inability of long-day photoperiods to terminate diapause if the temperature is too low (Schechter et al. 1971).

The most effective wavelength for inducing the termination of diapause of European corn borer larvae has not as yet been determined. However, in the codling moth (<u>Laspeyresia pomonella</u>) larvae and the oak silkworm (<u>Antheraea pernyi</u>) pupae, light in the blue region of the spectrum was shown to be most effective in breaking diapause (Norris et al. 1969; Schechter et al. 1971). Schechter et al. suggested, therefore, that a pigment or pigment-protein

which absorbs in this region is involved in triggering the entire series of biochemical events responsible for diapause termination. Hemes, bile pigments, and carotenoids are possible candidates for this pigment.

THE "PHOTOPERIODIC CLOCK"--CIRCADIAN OSCILLATOR VERSUS AN HOURGLASS MECHANISM

Since light via photoperiod plays such an important role in the initiation and termination of diapause, insects must have evolved some mechanism for recording the hours of light and/or darkness. Such a mechanism has been dubbed a "photoperiodic" or "biological clock". Currently there exists a controversy concerning the appropriateness of using the term "biological clock" when referring to this Those who feel that the term is a misnomer mechanism. explain that "clock" connotes a single discrete organ or system, and this concept is probably incorrect. As Beck (1968) and Brady (1974) have hypothesized, the timemeasuring capability is the result of an integration of numerous rhythmic functions, a combination of an unknown number of biochemical events. But, Brady has also pointed out that the term "biological clock" is a necessary term for the sake of convenience; and its use certainly is appropriate, as long as it is operationally defined as above.

Both circadian rhythms and hour glass mechanisms have been implicated in the functioning of the "biological clock" which controls the initiation and termination of insect diapause. According to Halberg (1959), a circadian rhythm is defined as a biological rhythm whose period is an approximation of the period of the earth's rotation. It is a free-running rhythm, an endogenous oscillator with a period of approximately 24 hours. Circadian rhythms can be entrained by the 24 hour light-dark cycle. When entrained, they can have a period of exactly 24 hours (Brady 1974). In circadian-based events, the underlying endogenous oscillation typically creates a period of greatest sensitivity for the production of some hypothetical metabolic product, the concentration of which controls the occurrence of a particular physiological event. In hour glass mechanisms, often referred to as interval timers, there is also an accumulation of some hypothetical metabolic product, and when this product reaches a given titre, the physiological event occurs. Photoperiod can control the occurrence of the physiological event by regulating the synthesis or lack of synthesis of the metabolic product through an alternation of light and dark reactions (Lees 1955, 1965); but, there is no underlying endogenous oscillation.

There is a whole class of circadian rhythms which involve developmental events occurring once in the life of

an individual, and which only appear as overt rhythms in synchronous populations. The term "gating" is applied to such events, implying that the event can only occur when the circadian clock opens the gate at the appropriate time (Brady 1974). Gated events can also operate via an hourglass mechanism (Brady 1974). In either case, if an individual misses a given day's gate, it must wait 24 hours for the next one. Truman (1972) has shown that in the tobacco hornworm, <u>Manduca sexta</u>, the endocrine events which initiate the molt (rather than the molts themselves) are gated events. Prothoracicotropic hormone (PTTH) was produced only during a gate in the early to middle portion of the night. Thus, those larvae which had completed development (were physically ready to molt) too late to exploit the gate waited until the next night to produce PTTH and molt.

Beck (1974, 1975, 1976 a,b) has proposed the "Dual System Theory" to explain the photoperiodic determination of diapause induction in the European corn borer. According to this theory an endogenous oscillator and an hour glass mechanism interact in determining whether the corn borer will diapause or complete its life cycle and pupate. At the onset of darkness, it is proposed that a substance (S) is rapidly produced, and reaches its maximum concentration within four hours. Then it is slowly metabolized away, the rate of degradation differing during the scotophase and the

photophase portions of the 24 hour period. Thus the concentration of S will vary depending upon the length of the scotophase. Under conditions of continuous darkness, the S-system is hypothesized to act as a free-running circadian rhythm. The concentration of S during a critical time span of the 24 hour period, a time span labelled the Determination Gate, determines whether the European corn borer will diapause or pupate. This determination gate is set by a second system, the P system, which Beck believes is entrained by the S system. P, too, is synthesized and reaches a maximum concentration within four hours. After this time it is slowly degraded. The Determination Gate coincides with that period of time when the concentration of P is between two arbitrary values. These values are fairly close together making the Determination Gate quite narrow. According to Beck, the "Dual System Theory" correctly predicts the diapause incidence under a variety of photoperiods including nondiel (photoperiods other than 24 hours) and skeleton photoperiods (photoperiods in which light breaks interrupt the scotophase).

Bowen and Skopik (1976), on the other hand, have purported that their findings support an hourglass mechanism for the "photoperiodic clock" which operates in the European corn borer. According to these researchers, circadian oscillations are not involved in the photoperiodic time measurement in this species. They based their conclusions on the inability of scotophases other than eight hours

(when coupled to sixteen hour photophases) to break larval diapause. They went on to explain that if a circadian rhythm were involved, a photoperiodic schedule consisting of 32 hours of dark and 16 hours of light should result in diapause termination. Then too, in their experiments 100% of diapausing corn borers exposed to a minimum of six 8-hour dark phases accompanied by at least 4 hours of light per cycle terminated diapause. Thus the total number of hours in each cycle appeared to be unimportant, a finding which is not consistent with a circadian mechanism. Their work, however, did not eliminate the possibility of diapause termination itself being dependent on an endogenous oscillator. Thus, while an hourglass may operate in part in creating the biochemical environment necessary to break diapause, the actual termination may be gated; and this gate may be controlled by a circadian rather than an hourglass mechanism.

While at first glance the conclusions of Beck seem to be at odds with those of Bowen and Skopik, it must be remembered that Beck's model was designed to explain diapause induction, while the experiments of Bowen and Skopik were concerned with diapause termination. It is possible that different mechanisms are operating in the two cases.

THE NEUROENDOCRINE CONTROL OF INSECT METAMORPHOSIS AND DEVELOPMENT

Nerve cells communicate with each other as well as with muscles, glands, and other organs by producing chemical messengers or neurosecretions. Basically, these are of two types, neurotransmitters and neurohormones. Researchers including Scharrer (1977, 1978) and Barker (1977) have attempted to provide definitions which can be used to distinguish between them. Thus, Barker (1977) has characterized neurotransmitters as "a low-molecular-weight class of molecules synthesized by nerve cells that underlie a form of cell-cell interaction in the nervous system characterized as synaptic transmission." Transmitters, then, mediate interactions between cellular elements, interactions which cover small discrete areas and which take place in very brief periods of time (milliseconds to seconds). Synaptic transmitters include the amino acids, glycine (gly), taurine (tau), proline (pro), and Y-amino butyric acid (GABA); the biogenic amines, norepinephrine, serotonin, and dopamine; and acetylcholine. Barker defines neurohormones as "substances released by nerve cells at extrasynaptic sites (portal, general or cerebrospinal circulations) to regulate the activity of both endocrine and nonendocrine target tissues." He goes on to explain that neurohormones differ from neurotransmitters because of their longer time of action, the location of their release sites, the increased

distance between release and target sites, their peptide structure, and their ability to cause multiple actions on a variety of neural and nonneural cellular elements. However, the distinction between the two classes of neurosecretions is not as clearcut as the above definitions would suggest; for as Barker (1977) points out, peptides have been found to be released in the central nervous system and to mediate short-term events via synaptic mechanisms as well as long-term events via hormonal mechanisms. His statement, "neurohormones may or may not be released at synapses and 'putative transmitters' with actions similar to peptides may be reconsidered as neurohormones," clearly exemplifies the state of confusion that currently exists in distinguishing between neurohormones and neurotransmitters. The problem seems to be in trying to tie structure (peptide or nonpeptide) to function/place of action (release into the general circulation or release at the synapse respectively). Perhaps it would be simpler to state the definitions in terms of structure or function and not both. This researcher would prefer to use function as the distinguishing characteristic and so define neurohormones as neurosecretions which exert their effects at target sites that are extrasynaptic and relatively far from their release sites. According to this definition a given molecule could act as both a neurotransmitter and a neurohormone. The underlying reason for

this duality of roles probably lies in the fact that before the evolution of endocrine glands, special neurosecretions served as regulatory hormones. With the evolution of additional endocrine apparatus (in arthropods and vertebrates), most of these hormones of neuronal origin now took on the role of mediators, integrating the functions of nervous and endocrine systems (Scharrer 1978).

Four endocrine and neuroendocrine glands (organs composed of nervous tissue which secrete neurohormones) have been shown to play important roles in controlling insect growth and metamorphosis. These are the neurosecretory cells of the brain, the corpora cardiaca and the corpora allata located in close proximity to the brain, and the prothoracic glands located in the prothorax and posterior parts of the head (Maslennikova 1973). Experimental evidence supports the following general sequence of humoral events in the regulation of growth and metamorphosis. The medial and lateral neurosecretory cells of the brain produce substances that are carried via nerves to the corpora cardiaca, one of which is believed to be PTTH (Scharrer 1952; Highnam 1958). The corpora cardiaca store and eventually release brain hormone (PTTH), a neurohormone into the blood. PTTH stimulates the release of \prec -ecdysone, the growth and molting hormone, from the prothoracic glands (Williams 1946; Williams and Adkisson 1964; Schneiderman and Gilbert 1964;

Yin and Chippendale 1975).¹ <-ecdysone is converted into B-ecdysone (MH), the true molting hormone by peripheral body tissues (Moriyama et al. 1970; King 1972a; Bollenbacher et al. 1975). B-ecdysone, then, stimulates other insect tissues such as the epidermis and gonads to initiate differentiation necessary for insect molting (Wigglesworth 1964; King 1972b; King et al. 1974). There is evidence indicating that B-ecdysone also acts as a positive feedback regulator stimulating certain neurosecretory cells of the brain's pars intercerebralis² to continue synthesizing and releasing PTTH (Agui and Kiyoshi 1977; Marks et al. 1972). JH³ secreted by the corpora allata influences the nature of the molt, whether it will be larval, pupal or adult. A brain hormone (an allatropin) is believed to stimulate the corpora allata to produce JH (Maslennikova 1973).

¹It is widely held that the brain produces several different hormones which modify the action of a variety of glands including the prothoracic glands and the corpora allatum. Ishizaki and Ichikawa (1967) separated three active prothoracicotropic fractions from extracts of silkactive prothoracicotropic fractions from extracts of silkactive prothoracicotropic fractions and Kobayashi (1969) worm (<u>Bombyx mori</u>) brains. Yamazaki and Kobayashi (1969) worm (Bombyx mori) brains. Yamazaki and corpora cardiaca. separated a fourth. Gersch et al. (1973) isolated two separate factors from cockroach brain and corpora cardiaca. It is significant that the effects of their two activators It is significant that the effects of their two activators were different. Activator I stimulated RNA synthesis while activator II increased membrane potential.

²The pars intercerebralis is the central portion of the insect brain where the medial neurosecretory cells are located.

³Many authors believe that the various effects of JH are due to the action of two or more different hormones secreted by the corpora allata (Maslennikova 1973; Yin and Chippendale 1973). Thus, insect growth and metamorphosis is primarily controlled by the balance between two hormonal factors, MH and JH. This balance between the two hormones is more important in controlling insect metamorphosis than is their absolute concentrations. JH has also been shown to stimulate ovogenesis in females and the development of accessory glands in males (Schneiderman and Gilbert 1959; Gilbert 1964; Wigglesworth 1966, 1970). It is interesting to note that neck-ligated tobacco hornworm larvae exhibit a delayed pupation even though PTTH cannot reach the prothoracic glands (due to the ligation) and induce them to produce \prec -ecdysone (Gibbs and Riddiford 1977). While "leaky" prothoracic glands have been held responsible for the larvalpupal molt in this instance, research confirming this hypothesis has not been forthcoming.

Wigglesworth (1934) was the first to suggest that hormones controlling insect growth and metamorphosis also played a role in diapause. Williams (1946) later showed that Cecropia moth (<u>Hyalophora cecropia</u>) diapause was initiated by the failure of the brain's neurosecretory cells to secrete a brain hormone, PTTH. A similar inactivity of the brain's neurosecretory cells was linked to the onset of diapause in the wheat stem sawfly (<u>Cephus cinctus</u>) (Church 1955) and the Lime hawk-moth (<u>Mimas tiliae</u>) (Highnam 1958). In the European corn borer, it appears that diapause is initiated by a block in the release of PTTH rather than by

its lack of production (Cloutier et al. 1962; McLeod and Beck 1963a). More recently, in insects exhibiting a larval diapause, evidence has been presented for the role of an intermediate titre of JH in arresting morphogenesis by inhibiting the transport or release of PTTH (Yagi and Akaike 1976; Chippendale and Yin 1973, 1976; Yin and Chippendale 1973, 1974, 1976). When JH titre has been reduced and PTTH is released into the blood in sufficient concentration to stimulate the prothoracic glands, diapause is terminated and pupation ensues. Takeda (1978), from electron micrograph studies, provided additional evidence for corpora allata control over PTTH production in the diapausing slug worm (Monema flavescens). He found a significant decrease in nuclear volume in neurosecretory B cells in the pars intercerebralis (associated with a build-up of neurosecretory material) with corpora allata extirpation.

Yin and Chippendale (1975) have confirmed that insect prothoracic glands (PTG) are inactive during diapause by comparing PTG ultrastructure in diapausing and non-diapausing Southwestern corn borer (<u>Diatraea grandiosella</u>) larvae. They found that in the former, the cells are bounded by a thin membrane, have large nuclei, and have glycogen particles evenly dispersed through the cytoplasm; while in the latter, the cells are bounded by a thick membrane, have shrunken convoluted nuclei, and have an inner cytoplasmic zone of glycogen particles. No mention was made concerning the time of day of these observations. It is possible that there are 24 hour rhythms in these structural changes.

That the mechanism of JH control over the neurosecretion of PTTH is somewhat different from that causing pupation in last instar larvae is shown by the fact that the cessation of JH production in non-diapausing Southwestern corn borers (Yin and Chippendale 1976) and tobacco hornworms (Nijhout 1975) does not immediately initiate the production of PTTH and pupation. In the Southwestern borer, JH production ceases within 24 hours after ecdysis to the fifth instar while pupation does not occur until later in this instar. In the tobacco hornworm, premature elimination of the source of JH (by extirpation of the corpora allata) in the final larval instar was not a sufficient stimulus to initiate pupation.

While generalizations concerning the role of hormones in controlling insect growth and metamorphosis have been made across insect orders, families, genuses, and species, such generalizations should be made with caution. As Judy (1974) points out, each species fits into a certain niche; and therefore, slight differences in metabolism and its control must be expected even in closely related species. Since the European corn borer was the experimental animal used in this study, it is appropriate at this time to discuss

the specific nature of its neuroendocrine structures and their control over corn borer development and metamorphosis.

Figure 8, from McLeod and Beck (1963b), provides a lateral view of the brain and other neuroendocrine organs in the head of the European corn borer larva. Their arrangement is similar to that found in other lepidopterans except for the fact that the corpus allatum and corpus cardiacum are fused into a single structure, the corpus allatum-cardiacum complex (CAC), whereas in most other lepidopterans they are separate (Cazal 1948). The rice stem borer, Chilo suppressalis, was also found to have a CAC (Mitsuhashi and Fukaya 1960). In the European corn borer, the CAC are located one on each side of the head just posterior to the bilobed brain. Each complex contains seven large secretory cells, and is connected to the brain by two nerves, the nervus corporis cardiaca I (NCC I) and the nervus corporis cardiaca II (NCC II). While McLeod and Beck observed neurosecretory material in NCC I, they have never seen any in NCC II.

As shown in Figure 9, from McLeod and Beck (1963b), the brain is composed of 40 neurosecretory cells which can be divided into three types according to location--lateral, medial, and posterior. Upon examining these neurosecretory cells in diapausing and mature non-diapausing larvae, McLeod and Beck found no differences in either size or staining properties. Neither did they note any effect of long day

Fig. 8. Lateral view of the brain and neuroendocrine organs of the European corn borer. NCC I, nervus corporis cardiaca I; NCC II, nervus corporis cardiaca II. (From McLeod and Beck 1963b).



Fig. 9. The neurosecretory cells of the larval brain of the European corn borer. LNC, lateral neurosecretory cells; MNC, medial neurosecretory cells; PNC, posterior neurosecretory cells. (From McLeod and Beck 1963b.)

48a



e e

0.

or short day photoperiods on neurosecretory cell structure or activity. Kono (1975), on the other hand, did find differences in the daily changes in activity of neurosecretory cells type II in short day and long day imported cabbageworm, <u>Pieris rapae crucivora</u>, larvae. (<u>Pieris</u> diapauses as a pupa.) He therefore concluded that photoperiodic time measurement of diapause induction in this insect depended upon the daily secretory cycle of neurosecretory type II cells which had been entrained by photoperiod during the larval stage.

Back in 1962, Cloutier et al. showed that in the European corn borer, diapause brains implanted into diapausing larval abdomens resulted in the termination of diapause. This finding led them to postulate the presence of a blood brain barrier in the diapausing larva. Thus, tissue damage during implantation upset the barrier and allowed the brain hormone (PTTH) to diffuse into the heamolymph. This in turn caused the termination of diapause. The results of Cloutier et al., then, offer a possible explanation for the lack of staining and size differences between the neurosecretory cells of diapausing and non-diapausing corn borer brains. Currently, it is held that the neurosecretory product necessary to stimulate the prothoracic glands and thus break diapause has already been produced before the induction of diapause, and that "a blood brain barrier" which prevents
its diffusion into the haemolymph is responsible for the onset of diapause. Houk and Beck (1975, 1976), based on an ultrastructural examination of diapause and non-diapause larval brains and horseradish peroxidase penetration studies, suggested that the perineurial type II¹ cells form the basis for the "blood-brain barrier," and that glial cells may also play an important role in the enzymatic regulation of this barrier. Their results supported those of McLaughlin (1974) and Eldefrawi et al. (1968), who carried out similar investigations on other insects.

In 1964, Beck and Alexander suggested that diapause development in the European corn borer was dependent on a physiological factor which they named proctodone. Proctodone was purported to be produced by epithelial cells of the anterior portion of the insect's hind gut. Furthermore, it was noted that the hormone's release was rhythmic, occurring only during the photophase of the photoperiod. However, Chippendale and Yin (1975) were unable to duplicate their results, and suggested instead, that the secretory activity of the ileum was due to lysosomal involvement in metamorphic reorganization. They did not, however, rule out a neural or neurosecretory role for the insect abdomen in the

¹ The insect brain is ensheathed by a non-cellular neural lamella. Subjacent to the neural lamella is a bilayered perineurium; the outermost layer consists of perineurial type I cells and the inner layer is composed of perineurial type II cells. Glial cells underlie the inner membrane of the perineurial type II cells.

regulation of growth and metamorphosis, since there is a good deal of evidence suggesting that neural signals originating in the abdomen influence the brain's initiation of metamorphosis. (Edwards 1966; Beck 1970). And while receptors in the head are believed to receive the photoperiodic and thermal signals which regulate the onset and termination of diapause, it is possible that proprioceptive input from the abdomen might supplement this exteroceptive input.

The role of JH in initiating, maintaining, and terminating larval diapause has been well-documented (Fukaya and Mitsuhashi 1961; Yagi and Fukaya 1974; Chippendale and Yin 1976). In 1976, Yagi and Akaike showed that the larval diapause of the European corn borer too was regulated by JH. They reported prolonging the larval period with applications of JH and of accelerating pupation by allectomizing diapausing larvae. Thus, they provided an explanation for Beck and Shane's (1969) observation that diapause larvae injected with relatively high doses of \prec - or β -ecdysone formed larval-pupal intermediates. While at first Beck and Shane's results appeared to contradict those of Cloutier et al. (1962), it is possible that the larvae used by the latter researchers were in late diapause (low JH titre) and, therefore, pupated upon receiving implants of diapausing brains rather than forming larval-pupal intermediates.

JH titres are, then, at an intermediate level during early diapause and taper off during late diapause. Chippendale and Yin (1975), based on their work with the Southwestern corn borer, have suggested that it is the brain which interprets environmental signals and controls the levels of JH accordingly. When JH titres are sufficiently low, the brain is induced to release PTTH which stimulates the prothoracic glands to release *<*-ecdysone and initiate pupation. Therefore the brain's secretion of neurohormones not only regulates JH titres, but is also regulated by them (in activating the prothoracic glands). It is probable that a similar mechanism operates in the European corn borer and other insects which diapause as larvae. JH's role in manipulating the "blood-brain barrier" is not yet known.

A ROLE FOR CYCLIC AMP IN INSECT GROWTH AND METAMORPHOSIS

There is evidence which suggests an important role for the cyclic AMP system as part of the regulatory mechanism controlling insect growth and metamorphosis. Castillon et al. (1976 a,b) found marked changes in adenyl cyclase activity and in cyclic AMP levels in homogenized Mediterranean fruit fly, <u>Ceratitis capitata</u>, as the insect progressed through the larval and pupal stages. De Reggi and Cailla

(1975) found that cyclic AMP levels varied greatly during the postembryonic development of the fruit fly, <u>Drosophila</u> <u>melanogaster</u>. The formation of the puparium, larval-pupal apolysis, and pupal-adult apolysis was marked by especially rapid and drastic rises in cyclic AMP titres. Cyclic AMP also appeared to be involved in the stimulation of the prothoracic gland to produce & ecdysone (Vedeckis et al. 1974, 1976), and in the action of bursicon in the hardening and darkening of the insect cuticle (Vandenberg and Mills 1974, 1975). Cyclic AMP phosphodiesterase activity, too, appears to vary considerably during insect development (Whitmore et al. 1973; Morishima 1973; Bielinska and Piechowska 1975; Catalan et al. 1975; Gelman and Hayes 1978). But, in most cases, the exact nature of the involvement of cyclic AMP in insect development is not known.

In vertebrate metabolism, however, the role of cyclic AMP in mediating hormone, neurohormone, and neurotransmitter action is fairly well understood due to the tremendous research effort in this area during the past decade. Since it has been suggested that cyclic AMP plays an important role in mediating these types of actions in insects and other invertebrates, it would be useful to discuss the mechanism of cyclic AMP action in vertebrate tissues.

Sutherland et al. (1965) originated the model in which cyclic AMP was portrayed as a second messenger in hormone

action. (See Figure 4.) Based on their earlier research, they concluded that cells of different organs contain receptor sites for a variety of hormones. As a result of hormone-receptor interaction, cyclic AMP was produced. This cyclic AMP, in turn, altered the rate of one or more cellular processes which resulted in an altered physiological response such as the production of a steroid or a second hormone.

As shown in Figure 10 from Weiss and Kidman (1969), adenyl cyclase is responsible for directing the synthesis of cyclic AMP from ATP (Sutherland et al. 1962). Cyclic AMP phosphodiesterase directs the breakdown of cyclic AMP to adenosine 5'-phosphate (5' AMP) (Butcher and Sutherland 1962). Thus the concentration of cyclic AMP at any given time is due to the relative activities of adenyl cyclase and cyclic AMP phosphodiesterase. Adenyl cyclase activity is totally particulate and localized in the cell membrane, whereas phosphodiesterase activity is largely soluble and occurs primarily in the cytoplasm (de Robertis et al. 1967; Weiss and Costa 1968b). However, Arch and Newsholme (1976) have pointed out that in invertebrates, two phosphodiesterases have been isolated, one with a low Km value and one with a high Km value. The low Km value diesterase was localized in the membrane while the high Km diesterase was found in the cytoplasm. While substances which increase the production of cyclic AMP may do so by stimulating adenyl cyclase

Fig. 10. The formation and breakdown of cyclic AMP. (From Weiss and Costa 1969.)

55a



activity and/or by inhibiting phosphodiesterase activity, the regulation of cyclic AMP concentration by physiological stimuli (e.g. hormones and neurotransmitters) is believed to occur mainly by influencing the activity of adenyl cyclase (Arch and Newsholme 1976).

Sutherland and his colleagues first discovered the importance of cyclic AMP as an intracellular mediator for the action of glucagon and epinephrine in the mammalian liver (Sutherland and Rall 1957; Rall et at. 1957). Since then cyclic AMP has also been implicated in the actions of adrenocorticotropic hormone (ACTH) (Haynes 1958), thyroid stimulating hormone (TSH) (Rall and Sutherland 1962), vasopressin (Handler et al. 1965), lutenizing hormone (LH) (Marsh et al. 1965), and many others. It is currently held that most, if not all, of the actions of cyclic AMP in mammalian systems are mediated in turn by the phosphorylation of proteins. Thus, the production of cyclic AMP activates a protein kinase. This activation results in the phosphorylation of a protein, which in turn, is directly responsible for the given physiological event.

In insects, too, cyclic AMP has been implicated in the action of several hormones. Vandenberg and Mills (1974, 1975) and Seligman and Doy (1972) have provided evidence which suggests a role for cyclic AMP in mediating the action of bursicon, a neurohormone necessary for cuticle

darkening and hardening. Filburn and Wyatt (1976) have reported measurable quantities of adenyl cyclase in Gecropia moth fat body, and have hypothesized that cyclic AMP is involved in enzyme regulation in a manner similar to that reported in vertebrate tissues. In Gecropia pupal wing epidermis, Applebaum and Gilbert (1972) have reported ecdysone-stimulated production of cyclic nucleotides. Berridge and Patel (1968) have linked the action of 5-hydroxytryptamine (5-HT) on insect salivary gland to cyclic AMP; and Prince et al. (1972) have purported that cyclic AMP is serving as an intracellular messenger in the action of this hormone. Cyclic AMP has also been shown to be involved in PTTH's tropic regulation of the production of ←-ecdysone by the prothoracic glands of the tobacco hornworm (Vedeckis et al. 1974, 1976).

Within the last decade the cyclic AMP system has also been shown to have an important role in the functioning of the vertebrate and invertebrate nervous systems (Weiss and Kidman 1969; Greengard et al. 1973). According to Greengard (1976) cyclic AMP is involved in the mediation of the postsynaptic actions of certain neurotransmitters, in the regulation of their synthesis, and in the functioning of microtubules of the vertebrate brain.

At the synapse, Greengard (1976) suggests a role for cyclic AMP in mediating the postsynaptic actions of some neurotransmitters through the cyclic nucleotide-dependent

phosphorylation of specific membrane proteins. Thus, the neutrotransmitter released from presynaptic terminals activates a neurotransmitter-sensitive adenylate cyclase present in the postsynaptic membrane. The cyclic AMP which is produced, activates a cyclic AMP-dependent protein kinase which catalyzes the phosphorylation of the substrate protein. This phosphorylated protein changes the permeability of the postsynaptic membrane which in turn causes a change in the "post synaptic potential." The "post synaptic potential" is terminated by a phosphodiesterase and a phosphoprotein phosphatase which hydrolyze the cyclic AMP to 5'-AMP and convert the substrate protein back to the non-phosphorylated form respectively. In insects, a link between cyclic AMP and phosphorylase has been demonstrated in the cockroach nerve cord (Robertson and Steele 1972).

In presynaptic terminals, cyclic AMP-dependent kinases have also been found to regulate the synthesis of neurotransmitters. Thus cyclic AMP has been shown to activate tyrosine hydroxylase which in turn regulates the production of dopamine and norepinephrine (Goldstein et al. 1973; Harris et al. 1974).

In vertebrate brain tissue, cyclic AMP (Breckenridge and Bray 1970) and protein kinase have been implicated in the functioning of the microtubule system. Microtubules

from chick brain have been found to contain a cyclic AMPdependent protein kinase which catalyzes the phosphorylation of a substrate protein (Sloboda et al. 1975). The relationship of this protein to microtubule function has not yet been determined.

The mechanism of action of cyclic AMP as a mediator, then, appears to be similar whether it is mediating hormone, neurohormone, or neurotransmitter action. In each case, cyclic AMP activates a protein kinase which catalyzes the phosphorylation of a substrate protein. This protein, in turn, is responsible for a given physiological response depending upon its place of action. While this mechanism cannot be labelled as universal, it is a good beginning assumption for all cyclic AMP-mediated metabolism.

Then too, whether involved in hormone, neurohormone, or neurotransmitter action, the production of cyclic AMP via adenyl cyclase activity has been found to be activator specific. For example, rat pineal gland adenyl cyclase is stimulated by 1-norepinephrine but is not stimulated by polypeptide hormones such as glucagon, ACTH or lutenizing hormone which enhance the activity of adenyl cyclase in other organs. Neither was it stimulated by the biogenic amines, histamine, d-norepinephrine, serotonin or dopamine (Weiss and Costa 1968a). Usually, a given tissue receptor is only sensitive to a specific activator as in the case of the pineal gland cited above. However, certain receptors have been found to be stimulated by more than one activator (Greengard et al. 1973), and a given activator has been shown to have an effect on several different receptors as, for example, the hormone norepinephrine on brain, pineal gland, hypophysis, salivary gland, ocular tissue, and oviduct or uterine tissue (Jost and Rickenberg 1971).

Fluoride appears to be an exceptionally potent non-Specific activator of adenyl cyclase activity; but, the mechanism responsible for this activation is not understood (Sutherland 1972; Maguire et al. 1975). It has been hypothesized that adenyl cyclase is held in the membrane in an inhibited state, and that perhaps, the hormones, neurotransmitters, and/or fluoride act by counteracting this inhibition (Schramm and Naim 1970; Perkins and Moore 1971; Sutherland 1972).

Rojakovick and March (1972) were the first to demonstrate the presence of the enzyme adenyl cyclase in insect nerve tissue, specifically in the brains of the Madagascar cockroach, <u>Gromphadorhina portentosa</u>. Norepinephrine and epinephrine as well as fluoride ion were found to activate enzyme activity, while ecdysterone (a β -ecdysone) inhibited this activity. Since then adenyl cyclase activity has also been reported in the cockroach thoracic ganglion (Greengard et al. 1973) and in the tobacco hornworm (<u>Manduca sexta</u>)

larval brain (Vedeckis and Gilbert 1973). In the thoracic ganglion, three separate adenyl cyclase receptor units sensitive to dopamine, octopamine, and serotonin respectively were found. A norepinephrine-sensitive receptor was not uncovered, and it was therefore concluded that this activator partially stimulated the three receptors already identified. In the tobacco hornworm brain, *S*-ecdysone only behaved as an activator in the presence of the fluoride ion. It should be noted that while $\mathscr{B} ext{-}ecdysone$ stimulated adenyl cyclase production by tobacco hornworm brains, it inhibited enzyme production by cockroach brains. Several explanations could be offered for these seemingly contradictory results. First, Rojakovick and March's experimental animals were in the adult stage while Vedeckis and Gilbert used larvae. Since the insect brain is known to exhibit a regulatory effect on several different glands, it is plausible to expect that cyclic AMP might serve as a mediator in several different metabolic pathways. It follows that a particular hormone might act as an activator in one pathway and an inhibitor in another, especially at different stages of insect development. Another possibility to explain the difference observed, is the absence of fluoride ion in the incubation vessel used to determine the effect of β -ecdysone on adenyl cyclase activity in cockroach brains. Finally, it must be remembered that different insects have evolved somewhat different mechanisms

for the biochemical regulation of growth and metamorphosis. Insects which exhibit different types of metamorphoses such as the cockroach, a hemimetabolous insect, and the tobacco hornworm, a holometabolous insect, would be even more likely to exhibit such differences than those having similar life cycles.

Another possible role for cyclic AMP in controlling insect growth and metamorphosis is its involvement in the conversion of environmental signals into biochemical ones. Rasenick et al. (1978) based on their results which suggest an association between the activation of secretory activity in the brains of diapausing Cecropia and oak silkworm pupae and transfer to long-day photoperiods, hypothesized that the increase in cyclic AMP represents a transduction of photoperiodic signals. They have localized this cyclic AMP "spike" in the median neurosecretory cells of the brain, cells which have been implicated as the source of PTTH. Houk and Beck (1977) have suggested that dopamine or an immediate metabolite may be involved in an endogenous timemeasuring system and/or diapause induction-termination of the European corn borer. In light of the numerous reports of a link between dopamine and its immediate metabolites and cyclic AMP, it is reasonable to anticipate that a role for one or more neurotransmitters and for cyclic AMP in such a time-measuring system might surface.

It should be pointed out that cyclic AMP has been implicated in the light-regulated circadian rhythms of serotonin, serotonin N-acetyltransferase, N-acetylserotonin, melatonin, and noradrenaline¹ in rat pineal gland (Axelrod 1974). During the night, levels of serotonin N-acetyltransferase, (Klein and Weller 1970), N-acetylserotonin (Brownstein et al. 1973) and melatonin (Lynch 1971) rise, while the level of serotonin falls (Quay 1963). These rhythms are generated by the increased production of the neurotransmitter noradrenaline during the scotophase portion of the 24 hour cycle (Klein and Berg 1970; Brownstein and Axelrod 1974). According to Axelrod (1974), the circadian rhythm of noradrenaline appears to arise from a "biological clock" located in the hypothalamus. This "clock" is entrained by environmental lighting (photoperiod). Noradrenaline stimulates the production of N-acetyltransferase which in turn catalyzes the conversion of serotonin to N-acetylserotonin (Weissbach et al. 1960, 1961); hence, the increase in N-acetylserotonin titres and the decrease in serotonin titres. The noradrenaline stimulated production of N-acetyltransferase is mediated by cyclic AMP (Klein and Berg 1970). Melatonin levels also increase at night because N-acetylserotonin is rapidly converted into melatonin. It is especially noteworthy that the adenyl

¹ The names norepinephrine and noradrenaline have both been used in referring to 3,4-Dihydroxyphenyl-ethanolamine HCl.

cyclase receptor becomes supersensitive to stimulation by noradrenaline at night, and yet the basal adenyl cyclase activity does not differ from that measured during the daylight hours (Romero and Axelrod 1974). This supersensitivity has been attributed to the decreased release of noradrenaline during the daytime. Since photoperiod is an important regulator of insect life cycles, it is possible that the intermediate steps between photoreception and the release of PTTH, or in the case of diapause the inhibition of the release of PTTH, might also involve both cyclic AMP and one or more of the biogenic amines, such as norepinephrine, dopamine, and octopamine.

Inasmuch as the timing mechanism is believed to be located in the insect head, it would be of interest to examine adenyl cyclase activity in larval heads of European Corn borers which have been exposed to LD 16:8 (pupa-inducing) and LD 10:14 (diapause-inducing) photoperiods. In addition, since norepinephrine appears to be involved in the lightregulated circadian rhythm controlling melatonin synthesis in the rat pineal, since dopamine and its immediate metabolites have been implicated in an endogenous time-measuring system and/or diapause induction-termination in the European corn borer, and since dopamine and octopamine have been found to stimulate adenyl cyclase activity in insect nerve tissue, it would also be of interest to determine the

effects of the neurotransmitters, norepinephrine, dopamine and octopamine on this activity.

CHAPTER III

MATERIALS AND METHODS

The procedures described herein were designed to determine adenyl cyclase levels in variously-aged fifth instar larval heads of European corn borers exposed to long day (LD 16:8) or short day (LD 10:14) photoperiods, and to determine the effects of the neurotransmitters, norepinephrine, dopamine, and octopamine on this activity. This chapter will discuss:

1. the methods used to culture the corn borer larvae

2. the methods used to identify the various stages of the fifth instar larvae

3. the adenyl cyclase assay

4. the method used to calculate adenyl cyclase activity in terms of moles of cyclic AMP produced per head per 30 minutes and in terms of moles of cyclic AMP produced per mg protein per 30 minutes

5. the methods used for the statistical analysis of the results.

MAINTAINING CORN BORER LARVAE

The European corn borers used in these investigations were obtained from the European Corn Borer Laboratory, Ankeny, Iowa. In Ankeny, on Wednesday of each week, egg masses were placed in Mason jars containing the medium of Lewis and Lynch (1969). Jars were shipped on Friday and were received at this researcher's laboratory on Wednesday or Thursday of the following week. At the time of arrival, larvae were usually in the second instar.

On the day of arrival, the jars were randomly divided into two groups. One group was placed in a biological oxygen demand box (BOD) under a light regimen of LD 16:8, pupation-inducing; the other group was placed in a similar BOD box under a light regimen of LD 10:14, diapause-inducing. The temperature was maintained at $24^{\circ} \pm 1.5^{\circ}$ C in both boxes.

IDENTIFICATION OF FIFTH INSTAR STAGES

Fifth instar larvae were identified by their head capsule width of 1.7-2.1 mm (Gelman and Hayes 1978). For the purposes of this study, LD 16:8 (LD) fifth instar larvae were classified as early (ELD), middle (MILD), mature (MLD), early prepupa (EPP), late prepupa (LPP), and pharate pupa (PHP); while LD 10:14 (SD) fifth instar

larvae were classified as early (ESD), middle (MISD), early mature (EMSD), late mature (LMSD), early diapause (EDIAP), late diapause (LDIAP), and refrigerated diapause (RDIAP). Although exposure to cold is not necessary for the termination of diapause (Beck 1963), it was decided to refrigerate diapausing larvae under a photoperiod of LD 10:14 because diapause termination is facilitated in such refrigerated larvae (personal observation). Early, middle, and mature fifth instar larvae were identified by their body diameter as follows:

Farly	1.0		1.5	mm	in	width	
Middle	2.0	+	0.3	mm	in	width	
Mature	3.0	+	0.3	mm	in	width	

(See Figure 11.) Early mature larvae were two to three weeks old; late mature larvae were four to five weeks old. Prepupae were identified by the whitish color of their gut contents (Folsom and Wardle 1934). This color resulted from the emptying of the gut in preparation for Pupation or diapause. Prepupae which had not enclosed themselves in silk were designated as early prepupae, while those which were encased in silk were designated as late prepupae. Pharate pupae were identified by their relative quiescence, light body color, reduced body turgidity, and sigmoid shape which was due to the forward curvature of the head into a somewhat tucked position (Hinton 1946).

Fig. 11. The early, middle, and mature fifth instar stages of the European corn borer.



LD 10:14 larvae which had not pupated by forty days of age and whose guts were white in color were considered to be in diapause. Larvae were considered to be in early diapause when they were 40 - 43 days of age, and in late diapause when they were 45 - 70 days of age. Early diapause larvae that had been placed in an LD 10:14 BOD box and kept at $4^{\circ} \pm 2^{\circ}$ C from the age of 40 days to the age of 70 - 84 days before being sacrificed were designated refrigerated diapause larvae.

THE ADENYL CYCLASE ASSAY

Adenyl cyclase activity of the various fifth larval instar heads was measured by a modification of the method of Krishna et al. (1968). The assay is designed to determine the amount of radioactive cyclic AMP produced from radioactive adenosine triphosphate (ATP) due to the action of adenyl cyclase. The breakdown of the cyclic AMP produced is prevented by the addition of the cyclic AMP phosphodiesterase inhibitor, theophylline.

Since preliminary experiments on LD mature larval heads revealed that adenyl cyclase activity was very low in the absence of the fluoride ion, sodium fluoride (NaF) was added to the incubation mixture. Adenyl cyclase levels

were determined for homogenates of larval heads of each of the thirteen groups described above. Based on the results of these experiments, it was decided to determine the effects of norepinephrine, dopamine, and octopamine on adenyl cyclase activity in head homogenates of seven of these groups: mature LD, late prepupae, pharate pupae, late mature SD, early diapause, late diapause, and refrigerated diapause larvae. Again NaF was added to the incubation mixture. Inasmuch as it has been reported that NaF masks the stimulatory effects of certain activators (Sutherland 1972), adenyl cyclase activity of mature LD, late prepupal, pharate pupal, early mature SD,¹ late mature SD, early diapause, late diapause, and refrigerated diapause larval heads was also determined in the absence of NaF.

As an added control procedure, the adenyl cyclase assay was performed on two mature LD larval head homogenates in the absence of the phosphodiesterase inhibitor, theophylline, and on two other mature LD head homogenates in the absence of theophylline and in the presence of 10 µl of a 0.2% 3',5'-cyclic AMP phosphodiesterase solution.² Results were compared with those of controls containing

After examining the effects of the three neurotransmitters on adenyl cyclase activity of head homogenates of the other stages, it was decided to test their effects on head homogenates of early mature SD larvae.

²The phosphodiesterase was purchased from Sigma Chemical Company. It was dissolved in Tris buffer, pH 7.4.

theophylline but no phosphodiesterase. This was done to insure that the radioactive cyclic AMP formed during the 30 minute incubation period was produced as a result of adenyl cyclase activity and not as a result of some other metabolic reaction.

To prepare each larval homogenate, three larvae were decapitated.¹ Their heads were homogenized by hand in 0.6 ml of cold incubation mixture (prechilled to 0°C by placing in ice) in a 2.0 ml glass homogenizer with glass pestle. The incubation mixture contained MgSO₄ (6.6 x 10^{-3} M), NaF (2 x 10^{-2} M), dithiothreitol (10^{-3} M), bovine serum albumin (fraction V) (0.01%) and theophylline when desired (2 x 10^{-2} M) dissolved in 4.0 x 10^{-2} M Tris-HCl (tris (hydroxymethyl) aminomethane HCl) buffer, pH 7.4.^{2,3} Two 0.1 ml aliquots of each homogenate were pipetted into 10 x 75 mm incubation tubes and allowed to come to room temperature. One of the tubes (the control) was then placed in a boiling water bath for 10 minutes. To determine the

All larvae were decapitated between the hours of 11 A.M. and 1 P.M.

²The results of preliminary experiments revealed that dithiothreitol and albumin were necessary for the expression of adenyl cyclase activity. These reagents appear to increase the stability of the enzyme.

³All the chemicals in the incubation mixture with the exception of the MgSO4 and the Tris were purchased from Sigma Chemical Company. The MgSO4 and the Tris were purchased from Fisher Chemical Company.

effect of norepinephrine, dopamine, or octopamine, $15 \mu l$ of a 0.05 M solution of the neurotransmitter or for comparison, of Tris buffer (pH 7.4) were added to the homogenizer before homogenization.

At intervals of 20 seconds, 0.1 ml of ATP (3.0 - 4.0 x 10^{-3} M) (¹⁴C - labeled ATP, specific activity 5 - 50 mc/mmol) was added to each incubation tube. The ATP was prepared by evaporating the solution containing the radioactive ¹⁴C ATP (purchased from New England Nuclear Corp.) to dryness, and then adding 5.0 ml of 4 x 10^{-3} M nonradioactive carrier ATP.² Evaporation was carried out by blowing a gentle stream of nitrogen gas over the solution of ¹⁴C ATP. The tubes were incubated for 30 minutes in a water bath at 34° ± 1°C. At the end of the 30 minute incubation period, 0.1 ml (0.5 mg) of a solution of carrier cyclic AMP² was added to each tube, and the reaction was stopped by placing the incubation tube in a boiling water bath for 3 minutes. The cyclic AMP was added so that the % recovery could be calculated following purification of the cyclic AMP. The cyclic AMP was isolated by chromatography on Dowex 50 - H^+ (Bio-Rad Laboratories, 50 W x 4 200 - 400 mesh) columns (0.4 by 3.3 cm) which were prepared

DL-norepinephrine HCl, DL-octopamine HCl, and dopamine were purchased from Sigma Chemical Company.

²The ATP and cyclic AMP were purchased from Sigma Chemical Company.

by pipetting 2.0 ml of a 50% (V/V) water suspension of the resin into columns and washing with distilled water. The 0.3 ml sample was carefully added to the column followed by a 1.7 ml distilled water wash. Elution was accomplished by carefully adding 2.0 ml portions of distilled water and collecting the second and third 2.0 ml fractions (which contained most of the cyclic AMP) in 15 ml conical centrifuge tubes. If the resin was disturbed by too forceful an addition, poor separation resulted. To the eluent were added 0.2 ml of Ba(OH)₂ (0.25 M) and 0.2 ml of ZnSO $_4$ (0.25 M) as described in Krishna et al. (1968). The tubes were centrifuged at 3500 rpm for 10 minutes. The Ba(OH) - ZnSO4 precipitation was repeated and the tubes were centrifuged at the same speed for 30 minutes. The supernatant was decanted and 2.0 ml of supernatant were added to 15.0 ml of Hydrofluor scintillation cocktail for counting in a Beckman liquid scintillation counter. Standards were prepared by adding 2.0 milliliters of appropriately diluted ^{14}C - labelled ATP to 15.0 ml of scintillation cocktail. Blanks were prepared by adding 2.0 ml of distilled water to 15 ml of scintillation cocktail.

Cyclic AMP recovery for each sample was calculated by comparing the moles of cyclic AMP in each supernatant with the moles of cyclic AMP in the 0.1 ml of solution added to each incubation tube just prior to boiling. The

absorption at 260 m μ of an aliquot of the desired solution or supernatant was used as a measure of the moles of cyclic AMP present.

CALCULATION OF ADENYL CYCLASE ACTIVITY

Moles of cyclic AMP produced per sample per 30 minutes were calculated as follows. The counts per minute (cpm) of each sample were corrected for % recovery by dividing total sample cpm by the % recovery for the sample. The corrected cpm for the control tube was then subtracted from that of the experimental tube. The number of moles of cyclic AMP produced was calculated from the equation:

Moles of ATP = Moles of ATP x cpm of unknown in unknown = in standard x cpm of standard cpm of blank

(The moles of ATP in the standard were ascertained by measuring its absorption at 260 m μ .) Multiplying the moles of ATP in the sample by the dilution factor, 4.8, yielded the number of moles of cyclic AMP produced per head per 30 minutes.

In order to negate the effects of changes in protein content of heads on adenyl cyclase levels, adenyl cyclase activity was also expressed in terms of moles of cyclic AMP produced per mg protein per 30 minutes. To assay for protein content of heads in any given group, one head was homogenized in 1.2 ml of 1.0 N sodium hydroxide. Duplicate 0.1 ml aliquots were pipetted, and protein concentration was determined by the method of Lowry et al. (1951). Four separate determinations were carried out for each of the thirteen larval stages. The number of moles of cyclic AMP produced per head per 30 minutes was then divided by the mg protein per head in order to determine the number of moles of cyclic AMP produced per mg protein per 30 minutes.

STATISTICAL ANALYSIS

To determine if there were any significant differences in adenyl cyclase activity in head homogenates of the thirteen stages tested, a thirteen group one way ANOVA (\prec = 0.05) was performed. Since the F ratio was significant, the Student Newman-Keuls procedure, as described in Winer (1962), was used to determine wherein the significant differences lay. A one way ANOVA (\checkmark = 0.05) was also used to ascertain if there were significant differences in the protein content of heads of the thirteen larval stages investigated. Since the F ratio was significant, the Student Newman-Keuls Procedure was used to determine wherein the significant differences lay.

A two way ANOVA ($\alpha = 0.05$) was used to determine if norepinephrine, dopamine and/or octopamine in the presence of NaF caused significant changes in adenyl cyclase activity in head homogenates of LD mature, late prepupal, pharate pupal, late SD mature, early diapause, late diapause and refrigerated diapause fifth instar larvae. A two way ANOVA (\varkappa = 0.05) was also employed to determine if in the absence of NaF any of the three neurotransmitters caused significant changes in adenyl cyclase activity in head homogenates of LD mature, late prepupal, pharate pupal, early SD mature, late SD mature, early diapause, late diapause and refrigerated diapause fifth instar larvae. Depending upon the significance of F for main effects and for interaction, the Student Newman-Keuls procedure was used to determine wherein the significant differences lay.

CHAPTER IV

RESULTS

COMPARISON OF ADENYL CYCLASE ACTIVITY IN HEAD EXTRACTS OF VARIOUS-AGED FIFTH INSTAR EUROPEAN CORN BORER LARVAE EXPOSED TO PHOTOPERIODS OF EITHER LD 16:8 (LD) OR LD 10:14 (SD)

Adenyl cyclase levels of homogenates of heads of the thirteen groups of fifth instar larvae described in Chapter III are compared in Figure 12. 1 Adenyl cyclase activity is expressed as moles of cyclic AMP produced per head per 30 minutes. Each mean represents the average of fifteen separate determinations. Using the error mean square as the best estimate of variance, the standard error for each group is \pm 1.0. The standard error for each group which has been indicated in Figure 12 has been calculated from the standard deviation of that group. As can be seen, in both LD and SD insect heads, adenyl cyclase activity in the presence of sodium fluoride increases as the larvae progress from early fifth through middle fifth to mature fifth instar. However, in LD heads, activity decreases as the larvae progress through the prepupal stage and is at its lowest level in the pharate pupa, whereas in SD heads, activity reaches a peak in early diapause and then in late diapause declines to the level exhibited by mature larval heads. The results of a one way ANOVA

In Table 9 of the Appendix, the individual values used in the determination of each group mean are listed.

Fig. 12. Adenyl cyclase activity in head extracts of LD 16:8 and LD 10:14 fifth instar European corn borer larvae expressed as moles/head/30 minutes. ELD, early long day; MILD, middle long day; MLD, mature long day; EPP, early prepupa; LPP, late prepupa; PHP, pharate EPP, early mature short day; MISD, middle short day; pupa; ESD, early short day; LMSD, late mature short EMSD, early mature short day; LMSD, late mature short emsp, early diapause; LDIAP, late diapause; RDIAP, day; EDIAP, early diapause. Stages having the same letters (within the bars) are not significantly different at the 0.05 level.

79a



MOLES/ HEAD/30 MIN. X 10-9

performed on the thirteen means are shown in Table 1. Since the F ratio was highly significant at the 0.05 level, the Student Newman-Keuls procedure was used to determine wherein the significant differences lay. In Figure 12, those means having the same letters are not significantly different at the 0.05 level. As can be seen, activity in early diapause heads is significantly higher than in all other groups and activity in late prepupal and pharate pupal heads is significantly lower than in mature LD, mature SD, early prepupal, and diapausing larvae.

TABLE 1.--One way ANOVA summary table for the determination of significant differences in adenyl cyclase activity (expressed as moles/head/30 minutes) among stages of fifth instar LD 16:8 and LD 10:14 fifth instar larval heads

Source	Degrees of Freedom	Sum of Squares	Mean Square	F Ratio	F Probability
Treatment	12	6433.9	536.2	33.0	0.000
Error	182	2953.4	16.2		
	194	9387.3			

In order to learn if changes in protein concentration of heads paralleled changes in adenyl cyclase activity, the protein content of the heads of each of the thirteen groups studied was determined. The results are presented in Figure 13.¹ Each mean represents the average of four separate

In Table 10 of the Appendix, the individual values used in the determination of each group mean are listed.

Fig. 13. Milligrams of protein in heads of LD 16:8 and LD 10:14 fifth instar European corn borer larvae. ELD, early long day; MILD, middle long day; MLD, mature long day; EPP, early prepupa; LPP, late prepupa; PHP, pharate pupa; ESD, early short day; MISD, middle short day; EMSD, early mature short day; LMSD, late mature short day; EDIAP, early diapause; LDIAP, late diapause; RDIAP, refrigerated diapause. Stages having the same letters (within the bars) are not significantly different at the 0.05 level.



MG PROTEIN
determinations. The standard error for each group has been indicated. Using the error mean square as the best estimate of variance, the standard error for each group may be restated as \pm 0.02.

In LD heads, protein concentration appears at first to increase during the fifth instar, peaks in the mature larvae, and then decreases again and reaches a low in pharate pupae. In SD heads, protein concentration increases during the fifth instar and continues to increase reaching a peak in early diapause heads and then shows a slight decrease in late and refrigerated diapause heads. A one way ANOVA was performed on the protein concentration means of the thirteen groups studied. As shown by the results in Table 2, F was highly significant at the 0.05 level. Therefore, the Student Newman-Keuls procedure was used to determine wherein the significant differences lay. In Figure 13, those means having the same letters are not significantly different at the 0.05 level. It is important to note the exceptionally low values of the standard errors, for these low values are a reflection of the accuracy with which the insects were beheaded (the lower the variance, the greater the accuracy).

TABLE 2.--One way ANOVA summary table for the determination of significant differences in protein concentration among stages of fifth instar LD 16:8 and LD 10:14 fifth instar larval heads

And the second statement of th	An and a second s	and the state of t	and a second secon		
water deployee one of departs of the origin says. She have a specific the galaxies					
Source	Degrees of Freedom	Sum of Sq uar es	Mean Square	F Ratio	Probability
alana ang kang ng panagipinan ng pana sama na dipina dipana dipana dipana di	Name (Teams Design and Antonio Angel (California) and an angel and an angel and an angel and an angel and an a	0 22/1	0.019	21.136	0.000
Treatment	12	0.224	0.01)		
Error	39	0.035	0.001		
	51	0.259			a datan di kana

In Figure 14, adenyl cyclase activity for each of the thirteen groups is expressed as number of moles of cyclic AMP produced per mg protein per 30 minutes.¹ The standard error for each group has been indicated. Using the error mean square as the best estimate of variance, the standard error for each group may be restated as [±] 1.7. Since a one way ANOVA yielded a significant F value at the 0.05 level (see Table 3), the Student Newman-Keuls procedure was again utilized to determine wherein the significant differences lay. In Figure 14, those means having the same letters are not significantly different at the 0.05 level. Overall results are almost identical to those shown in Figure 12 except for the refrigerated diapause group which is no longer significantly different from the

In Table 11 of the Appendix, the individual values used in the determination of each group mean are listed.

Fig. 14. Adenyl cyclase activity in head extracts of LD 16:8 and LD 10:14 fifth instar European corn borer larvae expressed as moles/milligram protein/30 minutes. ELD, early long day; MILD, middle long day; MLD, mature Hong day; EPP, early prepupa; LPP, late prepupa; PHP, long day; ESD, early short day; MISD, middle short pharate pupa; ESD, early short day; LMSD, late mature day; EMSD, early mature short day; LMSD, late mature short day; EDIAP, early diapause; LDIAP, late diapause; short day; EDIAP, early diapause. Stages having the same RDIAP, refrigerated diapause. Stages having the same letters (within the bars) are not significantly different at the 0.05 level.



middle SD group. This similarity indicates that accompanying differences in protein concentration of heads are not sufficient to explain the observed changes in adenyl cyclase activity.

TABLE 3.--One way ANOVA summary table for the determination of significant differences in adenyl cyclase activity (expressed as moles/head/milligram protein) among stages of fifth instar LD 16:8 and LD 10:14 fifth instar larval heads

Source	Degrees of Freedom	Sum of Squares	Mean Square	F Ratio	F Probability
Treatment	12	11847.4	987.3	22.7	0.000
Error	182	7900.8	43.4		
	194	19748.2			an an the state of

In order to insure that the radioactive cyclic AMP formed during the 30 minute incubation period was actually due to adenyl cyclase activity and not to some other metabolic reaction, adenyl cyclase activity in mature LD head extracts was also determined in the absence of theophylline, and in the absence of theophylline and the presence of phosphodiesterase. The results are presented in Table 4. As can be seen, the concentration of radioactive cyclic AMP after incubation is considerably reduced when the phosphodiesterase inhibitor is omitted, and is further reduced to only 3.0×10^{-9} moles per head per 30 minutes in the presence of added phosphodiesterase. Thus, it is apparent that the TABLE 4.--Adenyl cyclase activity of head extracts of mature LD 16:8 fifth instar larvae in the absence of theophylline and in the absence of theophylline and the presence of phosphodiesterase

		(Moles/Head/30 Minutes)	
	Theophylline (Contrgl) x 10-9	No Theophylline x 10-9	No Theophylline, Phosphodiesterase Added x 10-9
900000 (0000000000000000000000000000000	14.7	5.1	2.7
	9.7	6.1	3.4
Mean	12.2	5.6	3.1

EFFECTS OF NOREPINEPHRINE, DOPAMINE, AND OCTOPAMINE ON ADENYL CYCLASE ACTIVITY IN HEADS OF MATURE LD, LATE PREPUPAE, PHARATE PUPAE, LATE MATURE SD, EARLY DIAPAUSE, LATE DIAPAUSE, AND REFRIGERATED DIAPAUSE FIFTH INSTAR LARVAE IN THE PRESENCE OF SODIUM FLUORIDE

Since adenyl cyclase activity peaked in early diapause heads and decreased markedly in late prepupal and pharate pupal heads but not in late diapause heads, it was decided to investigate the effects of norepinephrine, dopamine, and octopamine on head extracts of the seven groups specified. The inability of these neurotransmitters to influence adenyl cyclase activity in the presence of sodium fluoride is shown by the results in Table 5.¹ The mean for each of the twenty-eight groups (seven stages x four activators)² and its standard error are indicated. Each mean represents the average of nine separate determinations. Using the error mean square as the best estimate of variance, the standard error for each group (cell mean) may be restated as \pm 1.9. The results of a two way ANOVA are shown in Table 6. Only the main effect due to stages was significant at the 0.05 level. Based on the results shown in Figures 12 and 14, it was expected that this main effect would be significant.

lIn Table 12 of the Appendix, the individual values used in the determination of each group (cell mean) are listed.

²For convenience, the term activator refers to the addition of norepinephrine, dopamine, octopamine or in the case of the control of Tris buffer to the incubation vessel. TABLE 5.--A summary of the effects of norepinephrine, dopamine, and octopamine on adenyl cyclase activity of seven stages of LD 16:8 and LD 10:14 fifth instar larval heads in the presence of sodium fluoride*,**

	LD Mature	Late Prepupa	Pharate Pupa	Late SD Mature	Early Diapause	Late Diapause	Refrig. Diapause
Control	13.7 ± 1.2	8.5 ± 0.9	2.5 ± 0.3	15.0 ± 1.0	35.5 ± 4.4	14.9 ± 1.5	13.2 ± 1.5
Norepinephrine	13.7 ± 1.0	8.9 ± 1.0	3.2 ± 0.5	17.3 ± 2.5	29.0 ± 3.2	15.5 ± 1.5	12.7 ± 1.2
Dopamine	10.8 ± 0.9	9.1 ± 0.7	2.7 ± 0.4	16.9 ± 2.7	31.8 ± 3.2	14.7 ± 1.1	14.0 ± 1.3
Octopamine	14.4 ± 0.7	8.4 ± 1.0	3.2 ± 0.4	19.1 ± 2.3	36.8 ± 4.8	13.9 ± 0.9	13.6 ± 1.4

*All values are multiplied by 10^{-9} and are expressed as moles of ATP converted to cyclic AMP per head per 30 minutes.

**The standard error for each group is stated after the group mean.

TABLE 6.--Two way ANOVA summary table for the determination of the presence of main effects and interaction in the analysis of the effects of norepinephrine, dopamine, and octopamine on adenyl cyclase activity of seven stages of LD 16:8 and LD 10:14 fifth instar larval heads in the presence of sodium fluoride

Source	Degrees of Freedom	Sum of Squares	Mean Square	F Ratio	F Probability
Stage	6	19058.5	3176.4	95.6	0.000
Activator	3	74.0	24.7	0.7	0.528
Interaction	18	442.3	24.6	0.7	0.768
Error	224	7446.4	33.2		
	251	27021.1			19 - 19 - 19 - 19 - 19 - 19 - 19 - 19 -

EFFECTS OF NOREPINEPHRINE, DOPAMINE, AND OCTOPAMINE ON ADENYL CYCLASE ACTIVITY IN HEAD EXTRACTS OF LD MATURE, LATE PREPUPAL, PHARATE PUPAL, EARLY SD, MATURE, LATE SD MATURE, EARLY DIAPAUSE, IATE MATURE, LATE SD MATURE, EARLY DIAPAUSE, IATE DIAPAUSE, AND REFRIGERATED DIAPAUSE FIFTH INSTAR LARVAE IN THE ABSENCE OF SODIUM FLUORIDE

Inasmuch as sodium fluoride (NaF) has been known to mask the effects of certain metabolites on adenyl cyclase activity, the effects of norepinephrine, dopamine, and octopamine on adenyl cyclase levels of head extracts of the eight stages listed above were also determined in the absence of NaF. Table 7 and Figure 15 show the results of

	LD Mature		Late Prepupa		pharate Pupa		Early SD	Mature	Late SD Mature	Jacce	Farly Diapause		Late Diapause		Refrig.	Diapause
Control	3.2±0	. 6	3.4±().5	3.3±	0.7	3.4±	0.4	2.8	1.1	5.5±	1.6	2.9±	0.6	3.6	±0.6
	А	a	A	а	А	а	А	а	А	a	А	а	А	а	А	a
Norepinephrine	2.1±0	.3	2.6±	0.5	3.2	0.3	6.7	1.5	4.8	±1.0	14.2-	£2.5	3.7-	0.9	3.8	3±0.8
	А	а	А	а	А	а	А	а	А	а	С	b	А	а	A	а
Dopamine	3.0+0).6	3.9±	0.7	3.6	±0.5	5.5	±0.7	5.2	±1.2	8.4	±2.0	1.94	0.7	6.5	5±0.6
	А	а	А	а	А	а	А	ab	А	ab	AB	Ь	А	а	А	ab
Octopamine	3.0+	0.6	5.41	.7	3.6	±0.7	7.2	±1.1	9.4	±2.1	11.5	±2.7	4.6	1.3	6.0	0 <u>+</u> 1.1
	А	а	А	ab	А	а	А	ab	В	bc	BC	С	А	а	A	ab

TABLE 7.--A summary of the effects of norepinephrine, dopamine, and octopamine on adenyl cyclase activity of eight stages of LD 16:8 and LD 10:14 fifth instar larval heads in the absence of sodium fluoride*,**

*All values are multiplied by 10⁻⁹ and are expressed as moles of ATP converted to cyclic AMP per head per 30 minutes. The standard error for each group is stated.

**Within each vertical column, those means having the same capital letter are not significantly different at the 0.05 level. Within each horizontal row, those means having the same lower case letter are not significantly different at the 0.05 level. Fig. 15. Pictoral representation of the effects of norepinephrine, dopamine, and octopamine on adenyl cyclase activity of eight stages of LD 16:8 and LD 10:14 fifth instar larval heads in the absence of sodium fluoride. C, control; D, dopamine; N, norepinephrine; O, octopamine; MLD, mature long day; LPP, late prepupa; PHP, pharate pupa; EMSD, early mature short day; LMSD, late mature short day; EDIAP, early diapause; LDIAP, late diapause; RDIAP, refrigerated diapause.



these experiments.¹ In Table 7, the mean for each of the thirty-two groups (eight stages x four activators) and its standard error are indicated. Each mean represents the average of six determinations. Using the error mean square as the best estimate of variance, the standard error for each group (cell mean) is \pm 1.1. Figure 15 is a pictoral representation of the effects of the three neurotransmitters on adenyl cyclase activity of head extracts of the eight stages investigated.

The results of a two way ANOVA are shown in Table 8. As can be seen, F values for both main effects and for interaction were significant at the 0.05 level. Therefore, the following procedure was used to investigate the nature of the interaction.

TABLE 8.--Two way ANOVA summary table for the determination of the presence of main effects and interaction in the analysis of the effects of norepinephrine, dopamine, and octopamine on adenyl cyclase activity of eight stages of LD 16:8 and LD 10:14 fifth instar larval heads in the absence of sodium fluoride

Source	Degrees of Freedom	Sum of Squares	Mean Square	F Ratio	F Probability
Stage	7	875.6	125.08	16.1	0.000
Activator	3	192.6	64.20	8.3	0.000
Interaction	21	343.3	16.35	2.1	0.005
Error	160	1244.7	7.78		
Total	191	1411.5	in the second	(Correct)	

In Table 13 of the Appendix, the individual values used in the determination of each group (cell mean) are listed. To determine if the effects of the neurotransmitters were significantly different from one another within any of the eight stages, the Student Newman-Keuls procedure was carried out on the means of each column in Table 7. Within any given column, those means having the same upper case letters are not significantly different at the 0.05 level. As can be seen, only in late SD mature and early diapausing larval head extracts is adenyl cyclase activity capable of being significantly stimulated by one or more of the neurotransmitters tested.

The Student Newman-Keuls procedure was also used to determine if there were any significant differences among the control means shown in the first horizontal row of Table 7. Although adenyl cyclase activity of head homogenates of early diapause larvae appeared to be somewhat higher than that of the other seven stages studied, it was not significantly higher than the rest at the 0.05 level.

Finally, in order to further analyze the interaction and to ascertain if, for a given neurotransmitter, there were significant differences when that neurotransmitter was acting in conjunction with a given stage, the Student Newman-Keuls procedure was conducted separately on the means of each of the last three horizontal rows of Table 7. For each neurotransmitter, those means having the same lower case letter are not significantly different from one

another at the 0.05 level. As can be seen, norepinephrine in conjunction with early diapause head extracts elicits significantly higher adenyl cyclase activity than norepinephrine in conjunction with any of the other seven stages studied. Similarly, dopamine in conjunction with early diapause head extracts elicits significantly higher adenyl cyclase activity than dopamine in conjunction with LD mature, late prepupal, pharate pupal or late diapause head extracts. It should be noted that dopamine in combination with early SD mature, late SD mature, and refrigerated diapause head extracts while not yielding significantly higher adenyl cyclase levels than in combination with any of the three LD 16:8 head extracts, does yield somewhat higher enzyme levels. Octopamine, too, exhibits a differential activation of adenyl cyclase depending upon the stage investigated. Thus, when octopamine is used in combination with early diapause head extracts, adenyl cyclase levels are significantly higher than when it is used in combination with head extracts from any of the other stages studied except for those from late SD mature larvae. When octopamine is used in combination with late SD mature head extracts, adenyl cyclase levels are significantly higher than when it is used with LD mature, pharate pupal or late diapause head extracts. Finally, octopamine in combination with late prepupal, early SD mature, and refrigerated diapause head extracts while not yielding significantly higher adenyl cyclase levels

than in combination with LD mature, pharate pupal, and late diapause head extracts, does result in somewhat higher enzyme levels.

SUMMARY OF RESULTS

In head extracts of fifth instar European corn borer larvae reared under both LD 16:8 and LD 10:14 photoperiods, adenyl cyclase activity (in the presence of NaF), expressed either as moles of cyclic AMP produced per head or per mg protein, increased as the larvae progressed through early, middle, and mature stages. In LD 16:8 larval heads, activity decreased in late prepupae and reached a low in pharate pupae. In contrast, in LD 10:14 larval heads, adenyl cyclase activity peaked in early diapause and then returned to prediapause levels during late diapause. When NaF was eliminated from the incubation mixture, there were no longer any significant differences in adenyl cyclase activity among head extracts of the various stages studied.

In the absence of NaF, norepinephrine significantly enhanced adenyl cyclase activity only in early diapause larval head extracts. Octopamine significantly enhanced adenyl cyclase activity in head extracts of late SD mature and early diapause larvae, but not in head extracts of the other six stages investigated. Dopamine did not significantly increase adenyl cyclase levels in head extracts of any of the eight stages investigated. In general, except for late diapause head extracts, a given neurotransmitter in combination with LD 10:14 larval head extracts resulted in higher adenyl cyclase levels than that neurotransmitter in combination with LD 16:8 head extracts. When sodium fluoride was present in the incubation vessel, none of the neurotransmitters enhanced adenyl cyclase activity.

CHAPTER V

DISCUSSION OF RESULTS AND RECOMMENDATIONS

In the presence of sodium fluoride (NaF), there were significant differences at the 0.05 level in the adenyl cyclase activity among early and late diapausing and various-aged fifth instar LD 16:8 (LD) and non-diapausing LD 10:14 (SD) larval heads of the European corn borer. (See Figures 12 and 14 and Tables 1 and 3.) Therefore, null hypothesis 1 which states that there is no difference in adenyl cyclase activity among early and late diapausing and various-aged fifth instar LD 16:8 (LD) and non-diapausing LD 10:14 (SD) larval heads of the European corn borer must be rejected. It should be stressed that the magnitude of the changes in adenyl cyclase activity of head extracts was not parallelled by similar magnitudes of change in the protein content of heads. This precludes the possibility that size variations alone were responsible for the observed differences in the adenyl cyclase levels of head extracts of the thirteen stages investigated. In the presence of NaF, adenyl cyclase levels in

early and middle LD larvae were significantly lower than those levels in mature LD and early prepupal larvae. Adenyl cyclase levels in early and middle SD larvae were also significantly lower than in late mature and all stages of

diapausing larvae. Thus, under both LD and SD conditions, adenyl cyclase activity was low at the beginning of the instar. In the prothoracic glands of <u>Manduca sexta</u>, adenyl cyclase levels were also reported to be relatively low during the early part of the fifth instar. (Vedeckis et al. 1976).

However, after the fifth instar larva has reached the mature stage, the photoperiodic regimen appeared to be quite important in influencing the levels of adenyl cyclase activity. Thus, in LD larval heads, adenyl cyclase activity was significantly lower in the late prepupa and pharate pupa than in the mature and the early prepupa. This indicates that cyclic AMP while probably having a role in the stimulation of PTTH and the initiation of pupation (Rasenick et al. 1976), may not be involved in pupation itself. In fact, it appears that the maintenance of higher levels of cyclic AMP in the mature larva may be in some way inhibitory toward the formation of the pupa. Catalan et al. (1976b) found that in whole homogenates of the fly, Ceratitis capitata, cyclic AMP levels which had reached a maximum during the third instar decreased towards the end of larval development and reached a minimum during apolysis (loosening of the exoskeleton prior to the larval-pupal molt). While there was not a similar decline in adenyl cyclase activity in <u>Ceratitis</u> capitata (Catalan et al. 1976a), there was a significant decrease in adenyl cyclase levels towards the

end of larval development and the onset of apolysis in European corn borer head homogenates. Catalan et al. (1976b) also found that cyclic GMP rose markedly just after apolysis and remained at a high level during the pupal stage. Therefore, they concluded that in the fly cyclic AMP plays a more important role than cyclic GMP in larval development and puparium formation while cyclic GMP rather than cyclic AMP mediates pupal development. It seems likely this is also the case in the European corn borer.

In corn borers exposed to SD photoperiods, adenyl cyclase activity peaked in early diapause heads. Since sodium fluoride was not capable of activating adenyl cyclase levels in head homogenates of stages other than early diapause insects to those levels reached by these insects, there is probably a non-NaF stimulated component of adenyl cyclase activity in early diapausing head extracts which does not exist in head extracts of the other stages. This hypothesis is supported by the somewhat higher adenyl cyclase level exhibited by early diapausing larval head extracts in the absence of NaF. (See Table 5.) However, the data was not strong enough to support a statement of significant difference in this case.

The relatively high standard error shown in Figure 12 for adenyl cyclase activity in early diapause insect heads suggests that there is a very narrow peak of adenyl cyclase

activity at this stage. In order to determine the age at which this peak occurs, the adenyl cyclase assay should be performed on head extracts of homogeneous (in regard to age) populations of early diapause larvae. Future experiments are planned in which early SD fifth instar larvae will be isolated within a few hours after ecdysis. The adenyl cyclase assay will then be performed on head homogenates of early diapause larvae for each age ranging from 20 to 28 days. (Age in this case will be determined from the time of ecdysis to the fifth instar.)

The fact that adenyl cyclase activity peaked in head extracts of early diapause larvae indicates that cyclic AMP is involved in some way in the onset of diapause. The exact nature of this involvement is open to speculation. Cyclic AMP might be involved in some way in the brain's activation of the corpora allata. Perhaps, in early diapause larval heads, the relatively high titres of adenyl cyclase result in increased levels of cyclic AMP which could be stimulatory to an as yet undetected "diapause hormone," the identity of which might be an allotropin. An allotropin is a hormone, typically secreted by the brain, which stimulates the corpora allata to produce JH. As was discussed earlier, JH levels appear to be maintained at an intermediate level in diapausing larvae, and these intermediate levels of JH are believed to be responsible for the

maintenance of diapause. Since adenyl cyclase activity in late and refrigerated diapause insect heads is significantly higher than in late prepupal and pharate pupal heads, it is possible that cyclic AMP might also play a role in maintaining this intermediate JH titre. Studies investigating adenyl cyclase levels in the corpora allatumcardiacum complex in early and late diapausing corn borer larvae would shed more light on this possibility.

Then too, cyclic AMP could be involved in the inhibition of the release of PTTH by the brains of diapausing insects. Thus the intermediate levels of JH in diapausing larvae could be stimulating the production of adenyl cyclase. The resultant cyclic AMP produced would then be activating a peptide which is responsible for maintaining the blood brain barrier and preventing the release of PTTH. Therefore, the adenyl cyclase activity of brains of diapausing larvae should be determined and compared with activity in brains of other fifth instar stages. The effect of JH on this activity should also be investigated.

In the absence of NaF, adenyl cyclase levels of head homogenates of prepupae and pharate pupae were no longer significantly lower than those of LD mature, SD early mature, SD late mature, late diapause, and refrigerated diapause larvae. (See Table 5.) These results are not contradictory to those shown in Figure 12, if as Perkins (1973) has noted, the addition of NaF to the incubation

mixture may provide a measure of the maximum level to which it is possible to stimulate adenyl cyclase activity in the sample. Thus, in the presence of NaF, the low adenyl cyclase levels in late prepupal and pharate pupal head extracts may indicate that adenyl cyclase activity can be stimulated only slightly or not at all during the late prepupal and pharate pupal stages respectively, and further supports the idea that cyclic AMP may not play an important role in European corn borer pupation.

In the presence of NaF, null hypotheses 2, 3, and 4 which state:

2. There is no effect of norepinephrine, dopamine or octopamine on adenyl cyclase activity in selected stages of LD 16:8 (LD) and LD 10:14 (SD) fifth instar larvae

3. There is no difference in the effects of norepinephrine, dopamine and octopamine on adenyl cyclase activity within any one of the fifth instar stages selected

4. There is no difference in the effect of any one of the neurotransmitters on adenyl cyclase activity among the fifth instar stages selected must be accepted at the 0.05 level. (See Tables 5 and 6.) However, in the absence of NaF, these null hypotheses must be rejected. (See Tables 7 and 8 and Figure 15.) Based on a comparison of the results in Tables 5 and 7, it is evident that NaF does mask the stimulatory effects of norepinephrine and octopamine on adenyl cyclase activity in head homogenates of early diapause larvae, and the stimulatory effect of octopamine on adenyl cyclase activity of head homogenates of late SD mature larvae.

It is especially significant that norepinephrine only enhanced adenyl cyclase activity in early diapause head extracts and octopamine only enhanced adenyl cyclase activity in late SD mature and early diapause head extracts, for this specificity of activation implies a differential sensitivity of the adenyl cyclase receptor units during the different stages investigated. Since such a differential sensitivity to norepinephrine has been implicated in the light-regulated circadian rhythms of those metabolites involved in melatonin synthesis in the rat pineal, it is possible that this differential sensitivity in corn borers exposed to SD and LD photoperiods might be involved in the light-regulated mechanism responsible for the onset of diapause. Houk and Beck (1977) have implicated dopamine or one of its immediate metabolites in the maintenance of the blood brain barrier which is believed to be responsible for preventing the release of PTTH from the insect brain and thus maintaining diapause. Could the role of dopamine or one of its metabolites (i.e. norepinephrine or octopamine) be to activate an adenyl cyclase receptor unit which in turn is more sensitive to this activation in insects exposed to SD than in insects exposed to LD photoperiods?

It is also noteworthy that dopamine and octopamine in

conjunction with head extracts of early SD mature and refrigerated diapause head extracts resulted in somewhat higher adenyl cyclase levels as compared to those neurotransmitters in the presence of LD mature, pharate pupae and late diapause head extracts. While not significantly higher, these increased adenyl cyclase levels suggest that there is not only a role for cyclic AMP in diapause-associated reactions in late SD mature and early diapause larvae, as discussed previously, but that there is also a role for cyclic AMP in diapause-associated reactions in early SD mature and refrigerated diapause larvae. Rasenick et al. (1978) found that brains of diapausing Cecropia and Antheraea pernyi pupae which had been transferred to long day photoperiods had a higher cyclic AMP content than brains of similar pupae left under short day conditions. This led them to conclude that cyclic AMP mobilizes brain hormone (PTTH) and is thus involved in the diapause termination of these insects. If cyclic AMP is involved in activating PTTH, this could explain the somewhat higher adenyl cyclase activity in head extracts of late prepupae in the presence of octopamine. It should be stressed that cyclic AMP is ubiquitous in

insect tissue; therefore, a particular activator functions only in stimulating the production of cyclic AMP in a limited number of given tissues. Thus, for example, in early diapause insects, norepinephrine might stimulate the production of cyclic AMP in a particular region of the brain, and since barriers to diffusion exist, in other cephalic structures too. Octopamine might stimulate cyclic AMP production in a different

brain region and/or in a particular area of the corpora allatum-cardiacum complex. Cyclic AMP may, therefore, be involved in two or more separate but complementary metabolic reactions or pathways related to the diapause condition. This again points to the need for investigating the effects of norepinephrine, dopamine and octopamine on adenyl cyclase activity in isolated brains and parts thereof, in corpora allata-cardiaca complexes, and in other insect structures. Since NaF did stimulate adenyl cyclase activity in all stages of mature LD, prepupal, mature SD, and diapause larval head extracts, it would be desirable to test the effects of neurohormones and neurotransmitters (other than those used in these investigations) on adenyl cyclase activity in head extracts, brain extracts and corpora allata-cardiaca complexes.

Then too, it would be valuable to perform some "around the clock" studies on the European corn borer in which adenyl cyclase levels were determined at different times in the photophase and scotophase portions of the 24 hour day, for it is certainly plausible to expect that differences in activity do occur during each 24 hour cycle. Rasenick et al., (1978) based on preliminary experiments, have reported the presence of such a diurnal rhythm in brain cyclic AMP levels in Cecropia and <u>Manduca sexta</u>. Testing the activating effects of norepinephrine, dopamine, octopamine, and other neurotransmitters, as well as of appropriate hormones, neurohormones, and other peptides during different times of the 24 hour day might also prove to be fruitful.

Finally in the area of insect control, the findings described herein should be considered in the design of future control methods. Since European corn borers in diapause are more resistant to control procedures than other stages of European corn borers, and since one reason for this resistance is their overall low metabolic rate, the relatively active cyclic AMP system might provide a vulnerable point for attack against European corn borer infestation. And the assay described herein, which is capable of measuring adenyl cyclase levels in insect tissue in picogram quantities, would be useful in determining the effects of potential insect regulators on this system. The results of the experiments described in this study support a link between adenyl cyclase activity and the onset, maintenance, and termination of diapause in the European corn borer. The nature of this link is open to speculation. Several possible mechanisms of action have been suggested, and the investigation of these mechanisms should provide several years of fruitful research for interested investigators.

A summary of the suggestions for future research

1. a determination of the exact time in the early diapausing larval head when adenyl cyclase activity peaks

2. a comparison of adenyl cyclase levels of brains of various stages of European corn borers exposed to LD and SD photoperiods

3. a comparison of adenyl cyclase levels of corpus allata-cardiaca complexes of various stages of European corn borers exposed to LD and SD photoperiods

4. determination of the effects of appropriate neurotransmitters, hormones, neurohormones, and other peptides on adenyl cyclase activity of brains and corpora allata-cardiaca complexes

5. a performance of "around the clock" studies on adenyl cyclase activity in head homogenates, brains, and corpora allata-cardiaca complexes of the European corn borer, and a determination of the effects of appropriate neurotransmitters, hormones, neurohormones, and other peptides on this activity

6. a determination of the effects of potential insect growth regulators on adenyl cyclase activity in appropriate tissues of the European corn borer and other insect pests. While the experiments reported in this study were directed solely at learning more about adenyl cyclase activity in various fifth instar stages of the European corn borer, it would be advantageous to learn if the cyclic AMP system is similarly linked to the diapause condition in other insects.

APPENDIX -- SUPPLEMENTARY TABLES

			LD 16:8			
	Early LD	Middle LD	Mature LD	Early Prepupa	late Prepupa	Pharate Pupa
	2.8 1.8 2.4 3.7 6.3 11.7 11.4 14.0 13.3 8.4 4.8 10.1 3.1 4.1 6.9	9.0 9.9 7.6 6.6 9.4 11.4 3.8 16.8 8.5 4.9 7.5 7.1 7.2 6.0 8.6	10.1 7.9 10.1 7.9 14.1 15.3 18.1 15.2 19.0 15.0 12.8 8.9 12.5 14.7 10.9	11.0 8.1 12.3 12.4 11.9 10.6 15.1 11.7 12.9 16.4 9.3 11.9 11.7 11.7 12.7	5.2 4.7 7.6 5.0 7.8 6.8 9.2 11.3 7.2 8.6 10.5 5.8 8.5	4.9 3.5 3.8 3.6 3.7 3.7 4.7 0.5 9.8 2.9 3.7 4.7 0.5 9.8 2.9 3.7 4.7 0.5 9.8 2.9 3.7 4.7 0.5 9.8 3.2 3.5 7.6 3.2 3.5 9.5 9.5 9.5 9.5 9.5 9.5 9.5 9.5 9.5 9
Mean	7.0±1.1	8.3±0.8	12.8±0.9	12.0±0.5	7.4±0.5	4.3+0.4

TABLE 9.--Adenyl cyclase activity (expressed as moles/head/30 minutes) in head extracts of LD 16:8 and LD 10:14 fifth instar larval heads*,**

			1	LD 10:14			
	Early SD	Middle SD	Early Mature SD	Late Mature SD	Early Diapause	Iate Diapause	Refrig. Diapause
	5.2 5.3 1.5 2.0 4.3 5.6 6.6 7.5 4.4 8.0 8.1 1.8 1.4 1.3 6.0	8.5 8.6 7.9 6.4 9.9 5.0 16.5 7.6 8.0 14.3 5.6 9.4 8.8 7.1 9.9	9.6 9.6 14.5 14.8 22.4 13.5 8.2 13.7 10.4 11.4 11.9 10.9 14.6 16.4 10.9	10.8 15.2 13.3 10.6 11.2 13.3 11.2 13.9 21.1 15.2 14.7 11.9 15.3 10.8 10.6	19.7 24.5 19.0 19.7 23.6 40.0 34.3 48.4 39.6 19.5 26.3 27.9 35.4 18.2 16.1	14.7 14.2 21.4 18.1 12.2 15.6 19.1 14.5 16.4 15.2 12.7 14.9 11.8 12.2 11.4	7.6 14.7 8.2 6.0 19.8 15.6 16.0 14.9 18.6 9.5 14.1 7.5 10.5 7.6 17.9
Mean	4.6±0.6	8.9±0.8	12.9±0.9	13.2±0.7	27.5+2.5	15.0±0.7	12.6±1.2

TABLE 9. -- Continued

*All values are multiplied by 10⁻⁹.

**The standard error for each group is stated after the group mean.

Stage	mg of protein	Stage	mg of protein	
Early LD	0.474 0.498 0.522 0.546	Early SD	0.516 0.450 0.516 0.474	
Mean	0.510±0.016	Mean	0.489±0.016	
Middle LD	0.468 0.492 0.592 0.548	Middle SD	0.492 0.576 0.558 0.570	
11/1	0.525±0.028	Mean	0.549±0.019	
Mature LD	0.690 0.624 0.612	Early Mature SD	0.600 0.624 0.630 0.678	
	0.624	Mean	0.633±0.016	
Mean Early Prepupa	0.642 0.600 0.620	Late Mature SD	0.654 0.642 0.660 0.654	
1.0	0.590	Mean	0.653±0.004	
Mean Late Prepupa	0.600 0.570 0.590 0.620	Early Diapause	0.670 0.670 0.670 0.670	
Mean	0.600±0.010	Mean	0.67±0.000	
Pharate Pupa	0.450 0.436 0.456	Late Diapause	0.642 0.642 0.612 0.612	
71.4	0.456+0.009	Mean	0.627±0.009	
Mean	0.450-0.009	Refrig. Diapause	0.630 0.624 0.576 0.576	
		Mean	0.601±0.015	

TABLE 10.--Milligrams of protein per head of LD 16:8 and LD 10:14 fifth instar European corn borer larvae*

*The standard error for each group is stated after the group mean.

				LD 16:8			
	Early LD		Middle LD	Mature LD	Early Prepupa	Late Prepupa	Pharate Pupa
	5.5 3.5 4.7 7.3 12.4 22.9 22.4 27.5 26.1 16.5 19.8 6.1 8.0 13.5 9.4		17.1 18.9 14.5 12.6 17.9 21.7 7.2 32.0 16.2 9.3 14.3 13.5 13.7 11.4 16.4	15.8 12.4 15.8 12.4 22.1 24.0 28.4 23.8 29.8 23.5 20.1 14.0 19.6 23.1 17.1	17.9 13.2 20.1 20.2 19.4 17.3 24.6 19.1 21.0 26.8 15.2 19.4 19.1 19.1 20.7	8.7 7.9 12.8 8.4 13.1 11.1 9.7 11.4 15.5 19.0 12.1 14.5 17.6 9.7 14.3	10.8 7.7 6.4 8.3 6.1 7.9 7.7 8.1 7.5 10.3 8.8 12.1 17.3 14.9 7.0
Mean	13.7±2	.1	15.8±1.5	20.1±1.4	19.5±3.3	12.4±3.3	9.4±3.2

TABLE 11.--Adenyl cyclase activity (expressed as moles/milligram protein/30 minutes) in LD 16:8 and LD 10:14 fifth instar larval heads***

		1	LD 10:14			
Early SD	Middle SD	Early Mature SD	Late Mature SD	Early Diapause	Late Diapause	Refrig. Diapause
$ \begin{array}{c} 10.6\\ 10.8\\ 3.1\\ 4.1\\ 8.8\\ 11.5\\ 13.5\\ 15.3\\ 9.0\\ 16.4\\ 16.6\\ 3.7\\ 2.9\\ 2.7\\ 12.3 \end{array} $	15.5 15.7 14.4 11.7 18.0 9.1 30.1 13.8 14.6 26.0 10.2 17.1 16.0 12.9 18.0	15.2 15.2 22.9 23.4 35.4 21.3 13.0 21.6 16.4 18.0 18.8 17.2 23.1 25.9 17.2	16.5 23.3 20.4 16.2 17.2 20.4 17.2 21.3 32.3 23.3 22.5 18.2 23.4 16.6 16.2	29.4 36.6 28.4 29.4 35.2 59.7 51.2 72.2 59.1 29.1 39.3 41.6 52.8 27.2 24.0	23.4 22.6 34.1 28.9 19.5 24.9 30.5 23.1 26.2 24.2 20.3 23.8 18.8 19.5 18.2	12.6 24.4 13.6 10.0 32.9 25.9 26.6 24.8 30.9 15.8 23.4 12.5 17.5 12.6 29.8
9.4±1.3	16.2±1.4	20.3±1.4	20.3±1.1	41.0±3.8	23.9±1.2	20.9+2.0

TABLE 11.--Continued

*All values are multiplied by 10^{-9} .

**The standard error for each group is stated after the group mean.

	Une problem		antenna franciska for an frankrik for an frankrik for an an frankrik for an an frankrik for an an frankrik for An frankrik frankrik frankrik for an an frankrik for an an frankrik for an	an Bernen bereiten Berlen aus der Kennen Anter verstehe Bernen von Berlen aus der Bernen Berlen aus der Bernen Im Bernen berlen aus der Bernen Anter Bernen Bernen Bernen Berlen aus der Berlen aus der Berlen aus der Berlen
	Control x 10-9	Norepinephrine x 10 ⁻⁹	Dopamine x 10 ⁻⁹	Octopamine x 10 ⁻⁹
	15.1 8.0 14.7	12.5 9.9 14.3 14.4	7.4 9.7 9.5 10.4	11.9 16.6 11.3 15.8
Mature LD	16.3 10.8 14.4 18.4 9.7	12.2 12.3 18.5 18.0 11.5	9.0 9.9 11.5 16.1 13.3	16.5 14.6 16.0 14.9 12.4
Mean	13.7±1.2	13.7±1.0	10.8±0.9	14.4±0.7
Late Prepupa	10.7 4.9 6.6 13.0 5.1 7.8 11.0 9.3 8.3	8.3 8.0 15.0 10.3 8.0 5.9 5.1 9.3 9.8	10.7 5.4 10.9 9.8 8.7 10.0 6.0 9.6 10.9	11.4 8.4 4.9 6.2 8.9 13.0 4.6 7.4 10.8
Mean	8.5±0.9	8.9±1.0	9.1±0.7	8.4±1.0
Pharate Pupa	3.6 1.5 2.3 1.8 2.2 2.6 3.3 1.4 3.6	2.2 1.4 4.1 4.8 4.9 4.9 1.8 3.1 1.7	2.7 3.6 3.7 2.6 1.3 1.4 1.1 3.4 4.1	2.3 2.4 2.2 3.1 3.2 5.1 4.3 2.0 4.0
Mean	2.5±0.3	3.2±0.5	2.7±0.4	3.2±0.4

TABLE 12.--Effects of norepinephrine, dopamine, and octopamine on adenyl cyclase activity of eight stages of LD 16:8 and LD 10:14 fifth instar larval heads in the presence of sodium fluoride*

*The standard error for each group is stated after the group mean.

ness belen dar, barrelen pros berreten Brannen Orre	Control x 10 ⁻⁹	Norepinephrine x 10-9	Dopamine x 10-9	Octopamine x 10-9
Late Mature SD	13.6 19.2 15.0 19.9 14.4 16.9 10.6 13.8 11.9	20.9 13.2 35.6 15.9 19.2 15.1 11.7 12.9 11.3	33.7 24.4 9.7 14.0 20.3 10.3 16.2 11.9 11.5	21.9 21.3 19.2 35.3 14.9 12.2 16.9 15.8 14.4
Mean	15.0±1.0	17.3±2.5	16.9±2.7	19.1±2.3
Early Diapause	52.5 47.3 34.9 47.4 46.4 27.7 25.7 17.9 19.7	31.1 41.2 37.3 29.8 38.7 28.5 24.0 15.2 14.8	26.9 50.6 28.6 41.5 34.4 31.9 30.2 18.7 23.3	43.0 51.8 58.3 34.5 46.9 28.3 32.0 18.0 18.1
Mean	35.5±4.4	29.0±3.2	36.8±3.2	36.8±4.8
Late Diapause	26.5 16.0 13.0 11.0 13.2 14.0 12.8 12.9 14.6	19.0 13.0 11.0 25.8 16.5 11.1 15.2 14.6 13.0	13.7 14.3 17.8 20.6 12.6 10.1 14.6 17.2 11.4	15.4 12.4 19.0 13.5 8.8 12.5 14.0 13.8 15.4
Mean	14.9±1.5	15.5±1.6	14.7±1.1	13.9±0.9
Refrig. Diapause	10.5 19.6 12.6 15.8 20.1 12.9 9.5 11.2 7.3	14.9 12.1 15.2 10.7 18.3 13.5 6.2 9.5 13.8	16.3 8.8 11.0 13.2 19.1 16.4 17.0 8.4 15.6	17.2 17.4 13.5 12.7 19.5 13.7 7.5 12.5 8.2
Mean	13.2±1.5	12.7±1.2	14±1.3	13.6±1.4

TABLE 12. -- Continued
0.		and the second		anta haran daram daram dara dara dara dara dara da
	Control x 10-9	Norepinephrine x 10-9	Dopamine x 10 ⁻⁹	Octopamine x 10 ⁻⁹
Mature LD	1.8 1.8 4.0 5.1 4.6	1.6 0.8 2.5 2.5 2.9 2.4	1.5 1.6 4.3 3.4 5.0 2.4	5.7 3.4 2.8 2.2 1.9 1.8
Mean	3.2±0.6	2.1±0.3	3.0±0.6	3.0±0.6
Late Prepupa	3.3 1.3 4.4 3.4 4.3 3.9	1.2 1.4 4.3 2.1 2.9 3.4	3.9 0.4 5.5 5.0 4.1 4.3	5.0 4.3 3.9 4.4 6.0 8.7
Mean	3.4±0.5	2.6±0.5	3.9±0.7	5.4±0.7
Pharate Pupa	5.6 1.9 4.0 3.6 3.7	4.4 3.0 2.9 3.3 3.3 2.1	3.9 5.4 3.4 3.7 3.9 1.3	2.6 6.4 2.6 4.9 3.4 1.3
Mean	3.3±0.7	3.2±0.3	3.6±0.5	3.6±0.7
Early Mature SD	4.3 3.2 4.2 4.2 2.7 2.0	3.5 7.0 4.5 9.7 12.3 3.4	6.6 6.0 3.8 2.8 6.9 7.0	8.3 4.4 8.5 6.0 4.7 11.1
Mean	3.4±0.4	6.7±1.5	5.5±0.7	7.2±1.1

TABLE 13.--Effects of norepinephrine, dopamine, and octopamine on adenyl cyclase activity of eight stages of LD 16:8 and LD 10:14 fifth instar larval heads in the absence of sodium fluoride*

$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	
Iate 0.0 3.1 2.6 5.1 1.2 2.5 6.8 5.1 Mature SD 3.4 7.7 4.8 3.1 2.6 2.1 15.0 7.6 5.4 10.2 9.1 Mean 2.8 ± 1.1 4.8 ± 1.0 5.2 ± 1.2 Mean 2.8 ± 1.1 4.8 ± 1.0 5.2 ± 1.2 9.1 11.0 20.4 7.6 5.4 10.2 9.1 11.0 20.4 7.6 5.4 10.2 9.1 11.0 20.4 7.1 18.5 2.9 1.2 9.1 11.6 1.9 21.1 4.8 7.1 15.6 1.9 11.6 1.9 9.1 11.0 16.2 12.3 9.1 11.0 16.2 14.2 ± 2.5 8.4 ± 2.0 11.4	amine)-9
Mean 2.8 ± 1.1 4.8 ± 1.0 5.2 ± 1.2 9.1 3.0 18.1 11.0 20.4 7.1 18.5 2.9 1.6 7.1 6.2 5.3 11.4 Diapause 5.9 21.1 4.8 1.9 11.6 15.6 11.2 12.3 9.1 11.0 16.2 Mean 5.5 ± 1.6 14.2 ± 2.5 8.4 ± 2.0	
Early Diapause 3.0 7.1 18.5 2.9 	+±2.1
Mean 5.5 ± 1.6 14.2 ± 2.5 8.4 ± 2.0 11.5	
	5±2.7
1.1 4.2 4.7 10.9 2.0 5.6 1.5 4.2 2.0 5.6 1.5 4.2 Diapause 4.1 1.5 1.4 2.6 4.8 0.4 0.3 2.6 3.5 6.3 0.4 3.6	
Mean 2.8±0.6 3.7±0.9 1.8±0.7 4.0	5±1.3
4.3 2.4 6.9 5.6 5.7 3.3 7.9 4.1 5.7 2.1 6.7 10.9 Diapause 3.4 7.6 4.7 4.7 1.7 3.1 7.9 6.7 3.9 3.6 4.1 5.0 3.9 3.9	5 4 7 7
Mean 3.6±0.6 3.8±0.8 6.5±0.6 6.0)±1.1

TABLE 13.--Continued

*The standard error for each group is stated after the group mean.

LIST OF ABBREVIATIONS

- AMP: adenosine monophosphate
- ANOVA: analysis of variance
- ATP: adenosine triphosphate
- CAC: corpus allatum-(corpus) cardiacum complex
- CPM: counts per minute
- CYCLIC AMP: adenosine 3',5'-cyclic monophosphate
- ELD: early long day
- EDIAP: early diapause
- EMSD: early mature short day
- EPP: early prepupa
- ESD: early short day
- GMP: guanosine monophosphate
- JH: juvenile hormone
- LD: long day
- LDIAP: late diapause
- SMSD: late mature short day
- LPP: late prepupa
- M: molar
- mc: millicurie(s)
- mg: milligram
- MH: molting hormone
- MILD: middle long day
- MISD: middle short day
- ml: milliliter

- MLD: mature long day
- mmol: millimole
- mu: millimicron
- M: micron
- ul: microliter
- N: normal
- NaF: sodium fluoride
- PHP: pharate pupa
- PTG: prothoracic glands
- PTTH: prothoracicotropic hormone
- RDIAP: refrigerated diapause
- RPM: revolutions per minute
- SD: short day

BIBLIOGRAPHY

- Adkisson P. L. (1965) Light-dark reactions involved in insect diapause. In <u>Circadian Clocks</u>, ed. J. Aschoff. Amsterdam: North Holland Pub. Co. pp. 344-350.
- Agricultural Research Service. (1975) USDA Coop. Econ.

Insect Rep. 25, 69-76.

- Agui N. and Kiyoshi H. (1977) Ecdysone as a feedback regulator for the neurosecretory brain cells in <u>Mamestra brassicae</u>. J. Insect Physiol. 23, 1393-1396.
- Andrewartha H G. (1952) Diapause in relation to the ecology of insects. <u>Biol. Rev.</u> 27, 50-107.
- Applebaum S. W. and Gilbert L. I. (1972) Stimulation of adenyl cyclase in pupal wing epidermis by β -ecdysone. <u>Dev. Biol.</u> 27, 165-175.
- Arch J. R. S. and Newsholme E. A. (1976) Activities and some properties of adenylate cyclase and phosphodiesterase in muscle, liver, and nervous tissues from vertebrates and invertebrates in relation to the control of the concentration of adenosine 3':5'-cyclic monophosphate. <u>Biochem. J.</u> 158, 603-622.
- Axelrod J. (1974) The pineal gland: a neurochemical transducer. <u>Science</u> 184, 1341-1348.
- Barker J. L. (1977) Physiological roles of peptides in the nervous system. In <u>Peptides in Neurobiology</u>, ed. H. Gainer. New York: Plenum Press. pp. 295-343.

Beck S. D. (1962a) Photoperiodic induction of diapause

in an insect. Biol. Bull. 122, 1-12.

- Beck S. D. (1962b) Temperature effects on insects: relation to photoperiodism. <u>Ent. Soc. Amer. No. Cent. Br. Proc.</u> 17, 18-19.
- Beck S. D. (1963) Physiology and ecology of photoperiodism. Ent. Soc. of Amer. 1, 8-16.
- Beck S. D. (1967) Water intake and the termination of diapause in the European corn borer, <u>Ostrinia nubilalis</u>. J. Insect Physiol. 13, 739-750.

Beck S. D. (1968) Insect Photoperiodism. New York: Academic Press.

Beck S. D. (1970) Neural and hormonal control of pupation in <u>Galleria mellonella</u> (Lepidoptera: Galleriidae).

Ann. Entomol. Soc. Amer. 63, 144-149.

- Beck S. D. (1974) Photoperiodic determination of insect development and diapause. I. Oscillators, hourglasses, and a determination model. <u>J. Comp. Physiol.</u> 90, 275-295.
- Beck S. D. (1975) Photoperiodic determination of insect development and diapause. III. Effects of nondiel photoperiods. <u>J. Comp. Physiol.</u> 103, 227-245.
- Beck S. D. (1976a) Photoperiodic determination of insect development and diapause. IV. Effects of skeleton photoperiods. <u>J. Comp. Physiol.</u> 105, 267-277.

- Beck S. D. (1976b) Photoperiodic determination of insect development and diapause. V. Diapause, circadian rhythms, and phase response curves, according to the dual system theory. J. Comp. Physiol. 107, 97-111.
- Beck S. D. and Alexander N. (1964) Chemically and photoperiodically induced diapause development in the European corn borer, Ostrinia nubilalis. Biol. Bull. 126, 175-184.
- Beck S. D. and Hanec W. (1960) Diapause in the European corn borer Pyrausta nubilalis (Hübner). J. Insect Physiol. 4, 304-318.
- Beck S. D. and Shane J. L. (1969) Effects of ecdysones on diapause in the European corn borer, Ostrinia nubilalis. J. Insect Physiol. 15, 721-730.
- Berridge M. J. and Patel N. G. (1968) Insect salivary glands: stimulation of fluid secretation by 5-hydroxytryptamine and adenosine 3'-5'-monophosphate.

Science 162, 462-463.

Bielinska M. and Piechowska M. J. (1975) Change in activity of phosphodiesterase of cyclic adenosine 3',5'-monophosphate and cyclic guanosine 3',5'-monophosphate during larval development of locust, Schistocera gregaria (Forsk.). Bull. Acad. Pol. Sci. Cl. II. Ser. Sci. Biol. 23, 1-6.

Bodenstein D. (1957) Importance of development of competence to respond to hormones. In <u>Recent Advances</u> <u>in Invertebrate Physiology</u>, ed. B. T. Scheer. University of Oregon Publications. pp. 197-211.

Bollenbacher W. E., Vedeckis W. V., Gilbert L. I. and O'Connor J. D. (1975) Ecdysone titres and prothoracic gland activity during larval pupal development of <u>Manduca sexta</u>. <u>Devel. Bio.</u> 44, 46-53.

- Bowen M. F. and Skopik S. D. (1976) Insect photoperiodism: The "T Experiment" as evidence for an hourglass mechanism. <u>Science</u> 192, 59-60.
- Brady J. (1974) The physiology of insect circadian rhythms. In <u>Advances in Insect Physiology</u>, Vol. 10, ed. J. E. Treherne, M. J. Berridge, and V. B. Wigglesworth. New York: Academic Press. pp. 1-115.

Breckenridge B. and Bray J. (1970) Cyclic AMP and nerve function. In <u>Advances in Biochemical Psychopharmacology</u>, Vol. 3, ed. P. Greengard and E. Costa. New York: Raven Press. pp. 325-333.

Brown J. J. and Chippendale G. M. (1977) Ultrastructure and respiration of the fat body of diapausing and nondiapausing larvae of the corn borer, <u>Diatraea</u> <u>grandiosella</u>. J. Insect Physiol. 23, 1135-1142.

Brownstein M. and Axelrod J. (1974) Pineal gland: 24 hour rhythm in norepinephrine turnover. <u>Science</u> 184, 163-165. Brownstein M., Saavedra J. M., and Axelrod J. (1973) Control of pineal N-acetylserotonin by a beta adrenergic receptor. Mol. Pharmacol. 9, 605-618. Butcher R. W. and Sutherland E. W. (1962) Adenosine 3'5'-phosphate in biological materials. I Purification and properties of cyclic 3'5'-nucleotide phosphodiesterase and use of this enzyme to characterize adenosine 3',5'-phosphate in human urine. J. Biol. Chem. 237, 1244-1250.

Castillon M. P., Catalan R. E., and Municio A. M. (1973) Adenyl cyclase variation during development of the insect <u>Ceratitis</u> capitata. FEBS Lett. 32, 113-115. Catalan R. E., Castillon M. P., and Municio A. M. (1976a) Cyclic nucleotide cyclase variation during development of the insect Ceratitis capitata. Biochem.

Biophys. Res. Commun. 69, 914-919.

- Catalan R. E., Castillon M. P., and Municio A. M. (1976b) Variation of the levels of cyclic AMP and cyclic GMP during development of the insect Ceratitis capitata. Biochem. Biophys. Res. Commun. 72, 184-189.
- Cazal P. (1948) Les glandes endocrines retro-cerebrales des insects. Bull. Biol. France et Belgique, Suppl. 32, 1-227.
- Chiarappa L. (1971) Crop Loss Assessment Methods FAO Manual on the Evaluation and Prevention of Losses by Pests, Disease and Weeds. Oxford: Alden Press.

Chino H., Sakurai S., Ohtaki T., Ikekawa N., Miyazaki H., Ishibashi M., and Abuki H. (1974) Biosynthesis of <-ecdysone by prothoracic glands in vitro. <u>Science</u>, Wash. 183, 529-530.

Chippendale G. M. (1977) Hormonal regulation of larval diapause. <u>Ann. Rev. Entomol.</u> 22, 121-138.

- Chippendale G. M. and Yin C. M. (1973) Endocrine activity retained in diapause insect larval. <u>Nature</u> 246, 511-513.
- Chippendale G. M. and Yin C-M. (1975) Reappraisal of proctodone involvement in the hormonal regulation of larval diapause. <u>Biol. Bull.</u> 149, 151-164.
- Chippendale G. M. and Yin C-M. (1976) Endocrine interactions controlling the larval diapause of the Southwestern corn borer, <u>Diatraea grandiosella</u>. <u>J. Insect Physiol</u>. 22, 989-995.

Church N. S. (1955) Hormones and termination and reinduction of diapause in <u>Cephus cinctus</u> Nort. (Hymenoptera:

Cephidae). <u>Canad. J. Zool.</u> 33, 339-369. Cloutier E. J., Beck S. D., McLeod D. G. R., and Silhacek

D. L. (1962) Neural transplants and insect diapause. <u>Nature</u> 135, 1222-1224.

De Reggi M. L. and Cailla H. L. (1975) Cyclic AMP levels in Drosphila during postembryonic development and in adults. <u>J. Insect Physiol.</u> 21, 1671-1674.

- De Robertis E., Rodriguez De Lores Arnaiz G., Alberici M., Butcher R. W., and Sutherland E. W. (1967) Subcellular distribution of adenyl cyclase and phosphodiesterase in rat brain cortex. J. Biol. Chem. 242, 3487-3493.
- De Wied D. and Gispen W. H. (1977) Behavioral effects of peptides. In Peptides in Neurobiology, ed. H. Gainer. New York: Plenum Press. pp. 397-448.

Edwards J. S. (1966) Neural control of metamorphosis in

Galleria mellonella. J. Insect Physiol. 12, 1423-1433.

Eldefrawi M. E., Toppozada A., Salpeter M. M., and O'Brien R. D. (1968) The location of penetration barriers in the ganglion of the American cockroach, Periplaneta

americana. J. Exp. Biol. 48, 325-338.

El-Minshawy A. M., Moussa M. E., and Hammad S. M. (1974) Anatomy and histology of the mature larva of Ostrinia nubilalis (Hbn.) (Lepidoptera, pyraustidae). Alex. J. Agric. Res. 22, 213-222.

Filburn C. R. and Wyatt G. R. (1976) Adenylate and guanylate cyclases of cecropia silkworm fat body. J. Insect

Physiol. 22, 1635-1640.

Folsom J. W. and Wardle R. A. (1934) Entomology with Special Reference to its Ecological Aspects. Phila .: Blakison.

- Fukaya M., and Mitsuhashi J. (1961) Larval diapause in the rice stem borer with special reference to its hormonal mechanism. Bull. Nat. Inst. Agr. Sci. (C) 13, 1-32.
- Gelman D. B., and Hayes D. K. (1978) Cyclic 3',5'-AMP phosphodiesterase activity in head extracts of the five larval instars and in extracts of brains of the fifth instar of the European corn borer, Ostrinia nubilalis (Hübner). J. Comp. Biochem. Physiol. In Press.
- Gersch M., Brauer R., and Birkenbeil H. (1973) Experimteffe untersuchungen zun wirkungsmechanismus der beiden entwicklungsphysiologisch aktiven fraktionen des "gehirnhormons" der insekten (aktivationsfaktor I und II) auf die prothoracaldruse. Experientia 29, 425-427.
- Gibbs D. and Riddiford L. M. (1977) Prothoracicotropic hormone in Manduca sexta: localization by a larval assay. J. Exp. Biol. 66, 255-266.

Gilbert L. I. (1964) Physiology of growth and development: endocrine aspects. In The Physiology of Insecta, ed. M. Rockstein. New York: Academic Press. pp. 149-225. Goldstein M., Anagnoste B., and Shirron C. (1973) The effect of trivastal, haloperidol and dibutyryl cyclic AMP on 14_{C} dopamine synthesis in rat striatum. J. Pharm. Pharmac. 25, 348-351.

Greengard P. (1976) Possible role for cyclic nucleotides and phosphorylated membrane proteins in postsynaptic actions of neurotransmitters. <u>Nature</u> 260, 101-108.

Greengard P., Nathanson J. A., and Kebabian J. W. (1973) Dopamine-, octopamine-, and serotonin-sensitive adenylate cyclases: possible receptors in aminergic neurotransmission. In <u>Frontiers in Catecholamine</u> Research. Great Britain: Pergamon Press. pp. 377-382.

Halberg F. (1959) Physiological 24-hour periodicity; general and procedural considerations with reference to the adrenal cycle. <u>Z. Vitamin-Hormon-Ferment</u>

forsch. 10, 225-296.

Halberg F., Johnson E. A., Nelson W., Runge W., and Sothern R. (1972) Autorhythmometry - procedures for physiologic self-measurements and their analysis. <u>The Physiology</u> <u>Teacher</u> 1, 1-11.

Handler J., Butcher R. W., Sutherland E. W., and Orloff J. (1965) The effect of vasopressin and of theophylline on the concentration of adenosine 3°5°-phosphate in the urinary bladder of the toad. J. Biol. Chem. 240, 4524-4526.

Harris J. E., Baldessarini R. J., Morgenroth V. H. III, and Roth R. H. (1974) Regulation of catecholamine synthesis in the rat brain in vitro by cyclic AMP. <u>Nature</u> 252, 156-158.

- Haynes R. C., Jr. (1958) The activation of adrenal phosphorylase by the adreno-cortcotropic hormone. J. Biol. Chem. 233, 1220-1222.
- Herman W. S. and Gilbert L. I. (1966) The neuroendocrine system of <u>Hyalophora cecropia</u> (L.) (Lepidoptera saturniidae). <u>Gen. Comp. Endocrinol.</u> 7, 275-291.
- Highnam K. C. (1958) Activity of the brain/corpora cardiaca system during pupal diapause 'break' in <u>Mimas tiliae</u> (Lepidoptera). <u>Quart. J. Microscop. Sci.</u> 99, 73-88.
- Hinton H. E. (1946) Concealed phases in the metamorphosis of insects. <u>Nature</u>, <u>London</u> 157, 552-553.
- Hong J. W. and Platt A. P. (1975) Critical photoperiod and daylength threshold differences between northern and southern populations of the butterfly, <u>Limenites</u> <u>archippus</u>. J. Insect Physiol. 21, 1159-1165.
- Houk E. J. and Beck S. D. (1975) Comparative ultrastructure and blood-brain barrier in diapause and nondiapause larvae of the European corn borer, <u>Ostrinia nubilalis</u> (Hübner). <u>Cell Tiss. Res.</u> 162, 499-510.
- Houk E. J. and Beck S. D. (1976) An enzymatic component of the insect blood-brain barrier: implications of DAB (3,3'-diaminobenzidne) oxidation. J. Insect Physiol. 22, 523-528.

Houk E. and Beck S. D. (1977) Distribution of putative neurotransmitters in the brain of the European corn borer, Ostrinia nubilalis. J. Insect Physiol. 23, 1209-1217.

Hyden H. (1969) Biochemical approaches to learning and memory. In The Alpbach Symposium 1968--Beyond Reductionism, ed. A. Koestler and J. R. Smythies.

New York: The MacMillan Co. pp. 85-117.

- Ishizaki H. and Ichikawa M. (1967) Purification of the brain hormone of the silkworm, Bombyx mori. Biol. Bull. 133, 355-368.
- Jost J. and Rickenberg H. V. (1971) Cyclic AMP. Ann. Rev. Biochem. 40, 741-774.

Judy K. J. (1974) Hormonal control of insect development. In Invertebrate Endocrinology and Hormonal Heterophylly,

ed. W. J. Burdette. New York: Springer-Verlag. pp.

King D. S. (1972a) Ecdysone metabolism in insects. Am. 7-28.

Zool. 12, 343-345.

- King D. S. (1972b) Metabolism of *d*-ecdysone and possible immediate precursors by insects in vivo and in vitro. Gen. Comp. Endocr. (Suppl.) 3, 221-227.
- King D. S. (1974) Photoperiodic induction and inheritance of diapause in <u>Pionea forficalis</u>. <u>Entomol. Exp. Appl.</u> 17, 397-409.

- King D. S., Bollenbacher W. E., Borst D. W., Vedeckis W. V., O'Conner J. D., Ittycheria P. I., and Gilbert L. I. (1974) The secretion of *A*-ecdysone by the prothoracic glands of <u>Manduca sexta</u> in vitro. <u>Proc.</u> <u>Nat. Acad. Sci.</u>, U.S.A. 71, 793-796.
- Klein D. C. and Berg G. R. (1970) Pineal gland: stimulation of melatonin production by norepinephrine involves cyclic AMP-mediated stimulation of Nacetyltransferase. In <u>Advances in Biochemical Psychopharmacology</u>, Vol. 3, ed. P. Greengard and E. Costa. New York: Raven Press. pp. 241-263.

Klein D. C. and Weller J. L. (1970) Indole metabolism in the pineal gland: a circadian rhythm in N-acetyl-

transferase. <u>Science</u> 169, 1093-1095. Kono Y. (1975) Daily changes of neurosecretory type-II cell structure of Pieris larvae entrained by short

and long days. <u>J. Insect Physiol.</u> 21, 249-264. Koukkari W. L., Duke S. H., Halberg F., and Lee J. (1974) Circadian Rhythmic Leaflet Movements: Student Exercise

in Chronobiology. <u>Chronobiologia</u> 1, 281-302.

Krishna G., Weiss B., and Brodie B. (1968) A simple sensitive method for the assay of adenyl cyclase.

J. Pharmacol. Exp. Therap. 163, 379-385.

Lees A. D. (1955) The physiology of diapause in arthropods. Cambridge: Cambridge University Press. Lees A. D. (1965) Is there a circadian component in the <u>Megoura</u> photoperiodic clock? In <u>Circadian Clocks</u>, ed. by J. Aschoff. Amsterdam: North Holland Publ. Co. pp. 351-356.

Lees A. D. (1968) Photoperiodism in insects. In <u>Photo-</u> physiology Vol. 4. New York: Academic Press. p. 47-137.

Lewis L. C. and Lynch R. E. (1969) Rearing the European corn borer, <u>Ostrinia nubilalis</u> (Hübner), on diets containing corn leaf and wheat germ. <u>Iowa St. J.</u> <u>Sci.</u> 44, 9-14.

Lowry O. H., Rosebrough N. J., Farr A. L., and Randal R. J. (1951) Protein measurement with the folin phenol reagent. <u>J. Biol. Chem.</u> 193, 265-283.

Lukefahr M. J. (1961) Factors related to the induction of diapause in the bink bollworm. PhD. Diss. A&M College of Texas.

Lynch H. J. (1971) Diurnal oscillations in pineal melatonin content. <u>Life Sciences</u> 10, 791-795.

Maguire M. E., Sturgill T. W., and Gilman A. G. (1975)

Frustration and adenylate cyclase. Metabolism 24,

287-299.
Marks E. P., Ittycheriah P. I., and Leloup A. M. (1972)
The effect of *B*-ecdysone on insect neurosecretion
in vitro. <u>J. Insect Physiol.</u> 18, 847-850.

Marsh J. M., Butcher R. W., Savard K., and Sutherland E. W. (1965) The stimulatory effect of lutenizing hormone on adenosine 3'5'-monophosphate accumulation in corpus luteum slices. J. Biol. Chem. 241, 5436-5440.

Maslennikova V. A. (1973) Hormonal control of insect diapause. In Humoral Control of Growth and Differentiation, Vol. II., ed. by J. Lobue and A. S. Gordon. New York: Academic Press. pp. 3-33.

McLaughlin B. J. (1974) Fine-structural changes in a lepidopteran nervous system during metamorphosis.

J. Cell Sci. 14, 369-387.

McLeod D. G. R. and Beck S. D. (1963a) Photoperiodic termination of diapause in an insect. Biol. Bull.

124, 84-96.

McLeod D. G. R. and Beck S. D. (1963b) The anatomy of the neuroendocrine complex of the European corn borer, Ostrinia nubilalis, and its relation to diapause. Ann. Entomol. Soc. 56, 723-727.

Minis D. H. (1965) Parallel peculiarities in the entrainment of a circadian rhythm and photoperiodic induction in the pink bollworm (Pectinophora gossypiella). In Circadian Clocks, ed. J. Aschoff. Amsterdam: North Holland Publishing Co. pp. 333-343.

Mitsuhashi J. and Fukaya M. (1960) The hormonal control of larval diapause in the rice stem borer, Chilo suppressalis. III Histological studies on the neurosecretory cells of the corpora allata during diapause and post diapause. Japanese Jour. Appl. Entomol. Zool. 4, 127-134.

Morishima I. (1973) Cyclic AMP phosphodiesterase activity during the development of the silkworm, Bombyx mori. J. Insect Physiol. 19, 2261-2265.

Moriyama H., Nakanishi K., King D. S., Okauchi T., Siddal J. B., and Haffere W. (1970) On the origin and metabolic fate of *d*-ecdysone in insects. <u>Gen. Comp.</u> Endocr. 15, 80-87.

Mutchmor J. A. and Beckel W. E. (1959) Some factors affecting diapause in the European corn borer, Ostrinia nubilalis (Hbn.). Canad. J. Zool. 37,

161-168.

Nathanson J. A. and Greengard P. (1973) Octopamine-sensitive adenylate cyclase: evidence for a biological role of octopamine in nervous tissue. <u>Science</u> 180, 308-310.

Nijhout H. F. (1975) Dynamics of juvenile hormone action in larvae of the tobacco hornworm, Manduca sexta (L.).

Biol. Bull. 149, 568-579.

Norris K. H., Howell F., Hayes D. K., Adler V. E., Sullivan W. N., and Schechter M. S. (1969) The action spectrum

for breaking diapause in the codling moth, <u>Laspeyresia</u> <u>pomonella</u> (L.) and the oak silkworm, <u>Antheraea pernyi</u> <u>guer</u>. <u>Proc. Natl. Acad. Sci.</u> 63, 1120-1127.

Parker H. L. and Thompson W. R. (1927) A contribution to the study of hibernation in the larva of the European corn borer (<u>Pyrausta nubilalis</u> hb). <u>Ann.</u> <u>Ent. Soc. Amer.</u> 21, 10-20.

Perkins J. P. and Moore M. M. (1971) Adenyl cyclase of rat cerebral cortex: activation by sodium fluoride and detergents. J. Biol. Chem. 246, 62-68.

Perkins J. P. (1973) Adenyl cyclase. In <u>Advances in Cyclic</u> <u>Nucleotide Research</u> Vol. 3, ed. P. Greengard and G. A. Robison. New York: Raven Press. pp. 1-64.

Prince W. T., Berridge M. J., and Rasmussen H. (1972)

- Role of calcium and adenosine-3':5'-cyclic monophosphate in controlling fly salivary gland secretion. <u>Proc. Nat. Acad. Sci.</u> U.S.A. 69, 553-557.
- Quay W. B. (1963) Circadian rhythm in rat pineal serotonin and its modifications by estrous cyclic and photoperiod. <u>Gen. Comp. Endocrinol.</u> 3, 473-479.
- Rall T. W. and Sutherland E. W. (1962) Adenyl cyclase. II. The enzymatically catalyzed formations of adenosine 3'5'-phosphate and inorganic pyrophosphate from adenosine triphosphate. J. Biol. Chem. 237, 1239-1245.

Rall T. W., Sutherland E. W., and Berthet J. (1957) The relationship of epinephrine and glucagon to liver phosphorylase. <u>J. Biol. Chem.</u> 224, 463-475.

Rasenick M. M., Neuburg M., and Berry S. J. (1976) Brain cyclic AMP levels and the initiation of adult development in the Cecropia silkworm. <u>J. Insect</u> <u>Physiol.</u> 22, 1453-1456.

Rasenick M. M., Neuburg M., and Berry S. J. (1978) Cyclic nucleotide activation of the silkmoth brain--cellular localization and further observations on the patterns of activation. J. Insect Physiol. 24, 137-139.

Robertson H. A. and Steele J. E. (1972) Activation of insect nerve cord phosphorylase by octopamine and adenosine 3',5'-monophosphate. J. Neurochem. 19, 1603-1606.

Rojakovick A. S. and March R. B. (1972) The activation and inhibition of adenyl cyclase from the brain of the Madagascar cockroach (<u>Gromphadorhina portentosa</u>). <u>Comp. Biochem. Physiol.</u> 43B, 209-215.

Romero J. A. and Axelrod J. (1974) Pineal B-adrenergic receptor: diurnal variation in sensitivity. <u>Science</u> 184, 1091-1092.

Ryder J. C. and Burbutis P. (1969) Systemic insecticides for control of European corn borer and green peach aphid on peppers. J. Econ. Entomol. 62, 1150-1151. Scharrer B. (1952) Intra-axonal transport in insects: role of corpus cardiacum: nerve inhibition of corpus allatum. Biol. Bull. 102, 261-272.

Scharrer B. (1977) Peptides in neurobiology: historical introduction. In Peptides in Neurobiology, ed. H.

Gainer. New York: Plenum Press. pp. 1-8.

Scharrer B. (1978) Peptidergic neurons: facts and trends.

Gen. Comp. Endocr. 34, 50-62.

Schechter M. S., Hayes D. K., and Sullivan W. N. (1971) Manipulation of photoperiod to control insects.

Israel J. of Entomol. 6, 143-166.

Schneiderman H. A. and Gilbert L. I. (1959) Substance with juvenile hormone activity among animals. In Cell Organism and Milieu, ed. D. Rudnick. New York: Ronald Press Co. pp. 157-187.

Schneiderman H. A. and Gilbert L. I. (1964) Control of growth and development in insects. Science 143,

325-333.

Schramm M. and Naim E. (1970) Adenyl cyclase of rat parotid gland: activation by fluoride and norepinephrine.

J. Biol. Chem. 245, 3225-3231.

Seligman I. M. and Doy F. A. (1972) Studies on cyclic AMP mediation of hormonally induced cytolysis of the alary hypodermal cells and of hormonally controlled DOPA synthesis in Lucilla cuprina. Israel J. Ent. 7, 129-142.

Shapiro I. D. and Pereverzev D. S. (1974) The resistance of corn to Ostrinia nubilalis Hbn. (Lepidoptera, Pyralidae) and the prospects for international cooperation on this problem. Entomol. Rev. 53,

158-161.

Sloboda R. D., Rudolph S. A., Rosenbaum J. L., and Greengard P. (1975) Cyclic AMP-dependent endogenous phosphorylation of a microtubule-associated protein. Proc. Natl. Acad. Sci. U.S.A. 72, 177-181.

Sutherland E. W. (1972) Studies on the mechanism of hormone action. Science 177, 401-408.

Sutherland E. W. and Rall T. W. (1957) The properties of an adenine ribonucleotide produced with cellular particles, ATP, Mg²⁺, and epinephrine or glucagon.

J. Am. Chem. Soc. 79, 3608-3614.

Sutherland E. W., Rall T. W., and Menon T. (1962) Adenyl cyclase. I. Distribution, preparation and properties.

J. Biol. Chem. 237, 1220-1227.

Sutherland E. W., Oye I., and Butcher R. W. (1965) The

action of epinephrine and the role of the adenyl cyclase system in hormone action. <u>Rec. Progr. Horm</u>

Res. 21, 623-646.

Sutherland E. W. and Robison G. A. (1966) The role of cyclic-3",5'-AMP in response to catecholamines and other hormones. Pharmacol. Rev. 18, 145-161.

Takeda N. (1978) Hormonal control of prepupal diapause in Monema flavescens (Lepidoptera). Gen. Comp.

Endocrinol. 34, 123-131.

Truman J. W. (1972) Physiology of insect rhythms. I. Circadian organization of the endocrine events underlying the moulting cycle of larval tobacco hornworms. J. Exp. Biol. 57, 805-820.

Truman J. W. and Riddiford L. M. (1974) Physiology of insect rhythms: III. the temporal organization of the endocrine events underlying pupation of the tobacco hornworm. J. Exp. Biol. 60, 371-382.

Vandenberg R. D. and Mills R. (1975) Adenyl cyclase in the haemocytes of the american cockroach. J. Insect Physiol. 21, 221-229.

Vedeckis W. V., Bollenbacher W. E., and Gilbert L. I. (1974) Cyclic AMP as a possible mediator of prothoracic gland activation. Zool. J. Physiol. Bd.

785, 440-448.

Vedeckis W. V., Bollenbacher W. E., and Gilbert L. I. (1976) Insect prothoracic glands: a role for cyclic AMP in the stimulation of \checkmark -ecdysone secretion. Mol. Cell. Endocrin. 5, 81-88.

Vedeckis W. V. and Gilbert L. I. (1973) Production of cyclic AMP and adenosine by the brain and prothoracic glands of Manduca sexta. J. Insect Physiol. 19, 2445-2457.

- Vinal S. C. and Caffrey D. J. (1919) The European corn borer and its control. <u>Mass. Agr. Sta. Bull.</u> 189, 1-71.
- Weiss B. and Costa E. (1968a) Selective stimulation of adenyl cyclase of rat pineal gland by pharmacologically active catecholamines. <u>J. Pharmacol.</u> <u>Exp. Therap.</u> 164, 310-319.
- Weiss B. and Costa E. (1968b) Regional and subcellular distribution of adenyl cyclase and 3'5'-cyclic nucleotide phosphodiesterase in brain and pineal gland. <u>Biochem. Pharmacol.</u> 17, 2107-2116.
- Weiss B. and Kidman A. D. (1969) Neurobiological significance of cyclic 3'5'-adenosine monophosphate. In <u>Advances in Biochemical Fsychopharmacology</u> Vol. I, ed. F. Greengard and E. Costa. New York: Raven Press. pp. 131-164.
- Weiss B. and Strada S. J. (1972) Neuroendocrine control of the cyclic AMP system of brain and pineal gland. In <u>Advances in Cyclic Nucleotide Research</u>, Vol. I, ed. P. Greengard and G. A. Robison. New York: Raven Press. pp. 357-374.

Weissbach H., Redfield B. G., and Axelrod J. (1960) Biosynthesis of melatonin: enzymatic conversion of serotonin to N-acetylserotonin. <u>Biochimica et</u> <u>Biophysica Acta.</u> 43, 352-355. Weissbach H., Redfield B. G., and Axelrod J. (1961) The enzymic acetylation of serotonin and other naturally occurring amines. <u>Biochimica et Biophysica Acta.</u> 54, 190-194.

Whitmore D., Applebaum S. W., and Gilbert S. I. (1973) Cyclic AMP phosphodiesterase activity in the midgut of Manduca sexta. J. Insect Physiol. 19, 349-354. Wigglesworth V. B. (1934) The physiology of ecdysis in Rhodmius prolixus (hemiptera). II Factors controlling moulting and metamorphosis. Quart J. Micr. Sci. 77, 191-223.

Wigglesworth V. B. (1964) The hormonal regulation of growth and reproduction in insects. In Advances in Insect Physiology, ed. J. W. L. Beament, J. E. Treherne, and V. B. Wigglesworth. New York: Academic Press. pp. 244-332.

Wigglesworth V. B. (1966) Hormonal regulation of differentiation and morphogenesis. In Cell

Differentiation and Morphogenesis. Amsterdam: North Holland Publ. Co. pp. 180-209.

Wigglesworth V. B. (1970) Insect Hormones. Edinburgh:

Oliver and Boyd.

Williams C. M. (1946) Physiology of insect diapause: The role of the brain in the production and termination of pupal dormancy in the giant silkworm, Platysamia cecropia. Biol. Bull. 90, 89-98.

- Williams C. M. (1969) Photoperiodism and the endocrine aspects of insect diapause. <u>Symp. Soc. Exp. Biol.</u> 23, 285-300.
- Williams C. M. and Adkisson P. L. (1964) Physiology of insect diapause in the oak silkworm, <u>Antheraea</u>

pernyi. Biol. Woods Hole 127, 511-525.

- Winer B. J. (1962) <u>Statistical Principles in Experimental</u> Design. New York: McGraw Hill.
- Yagi S. and Akaike N. (1976) Regulation of larval diapause by juvenile hormone in the European corn borer, <u>Ostrinia nubilalis</u>. J. <u>Insect Physiol</u>. 22, 389-392.
- Yagi S. and Fukaya M. (1974) Juvenile hormone as a key factor regulating larval diapause of the rice stem borer, <u>Chilo suppressalis</u>. <u>Appl. Ent. Zool.</u> 9, 247-255.
- Yamazaki M. and Kobayashi M. (1969) Purification of the proteinic brain hormone of the silkworm, <u>Bombyx mori</u>. <u>J. Insect Physiol.</u> 15, 1981-1990.

Vin C-M. and Chippendale G. M. (1973) Juvenile hormone regulation of the larval diapause of the Southwestern corn borer, <u>Diatraea grandiosella</u>. J. Insect Physiol. 19, 2403-2420.

Yin C-M. and Chippendale G. M. (1974) Juvenile hormone and the induction of larval polymorphism and diapause of the Southwestern corn borer, <u>Diatraea grandiosella</u>. J. <u>Insect Physiol.</u> 20, 1833-1847. Yin C-M. and Chippendale G. M. (1975) Insect prothoracic glands: function and ultrastructure in diapause and nondiapause larvae of <u>Diatraea grandiosella</u>. <u>Can.</u> J. <u>Zool.</u> 53, 124-131.

Yin C-M. and Chippendale G. M. (1976) Hormonal control of larval diapause and metamorphosis of the Southwestern corn borer, <u>Diatraea grandiosella</u>. J. <u>Exp.</u> <u>Biol.</u> 64, 303-310.